

Copyright Undertaking

This thesis is protected by copyright, with all rights reserved.

By reading and using the thesis, the reader understands and agrees to the following terms:

- 1. The reader will abide by the rules and legal ordinances governing copyright regarding the use of the thesis.
- 2. The reader will use the thesis for the purpose of research or private study only and not for distribution or further reproduction or any other purpose.
- 3. The reader agrees to indemnify and hold the University harmless from and against any loss, damage, cost, liability or expenses arising from copyright infringement or unauthorized usage.

If you have reasons to believe that any materials in this thesis are deemed not suitable to be distributed in this form, or a copyright owner having difficulty with the material being included in our database, please contact lbsys@polyu.edu.hk providing details. The Library will look into your claim and consider taking remedial action upon receipt of the written requests.

Pao Yue-kong Library, The Hong Kong Polytechnic University, Hung Hom, Kowloon, Hong Kong

http://www.lib.polyu.edu.hk



The Hong Kong Polytechnic University Department of Electrical Engineering

ELECTRICITY MARKET ANALYSIS WITH CO-EVOLUTIONARY COMPUTATION

By

ZHANG SHENG XIANG

A thesis submitted in partial fulfillment of the requirements for the Degree of Doctor of Philosophy

May 2008

Certificate of Originality

I hereby declare that this thesis is my own work and that, to the best of my knowledge and belief, it reproduces no material previously published or written, nor material that has been accepted for the award of any other degree or diploma, except where due acknowledgement has been made in the text.

ZHANG Sheng Xiang

ABSTRACT

The unique feature of the electricity power industry makes the power market more akin to an oligopoly. A great deal of work has been performed on analyzing power markets using the oligopoly models in the literatures. The Supply Function Equilibrium (SFE) and Cournot models have also been widely employed to model the electricity market. Many techniques, such as empirical analysis, agents-based simulation, iterative Nash Equilibrium (NE) search algorithms and complementarity program methods, have been utilized to determine the power market equilibrium. Despite of these previous efforts, no method is widely recognized as being an effective method for market equilibrium determination. Hence there is an urgent need to develop an effective and powerful approach for power market analysis and equilibrium determination.

Co-evolutionary computation is developed from traditional Evolutionary Algorithms (EAs), which simulates the co-evolutionary mechanism in nature and adopts the notion of ecosystem. It is a new methodology used to simulate the bidding behavior of the market players and to determine market equilibrium. The thesis applies co-evolutionary computation algorithms (CCAs) to solve the power market equilibrium and to study several important issues in power market analysis.

The first issue is that when transmission constraints are considered, the profit function of Generation Companies (GenCos) may be nonconcave and discontinuous with many local optima. Determination of market equilibrium becomes a more challenging task. A two-level evaluation process is developed in the thesis for the determination of the equilibrium. The Linear Supply Function Equilibrium (LSFE) and the Cournot market models are employed and the market equilibrium is determined by CCA. The existence or non-existence of the NE due to the transmission and generation capacity constraints are illustrated using a 2-bus test system and the IEEE 30-bus system. The effects of different parameters settings of the LSFE model on the equilibria are also studied and compared with those found based on the Cournot model.

The second issue examined is to determine the market equilibrium in a multiple pricing period. When compared to actual markets, earlier research works study only a single pricing period market. A multiple pricing period market with inter-temporal constraints should also be studied to make the simulation results comparable to actual market settings. The CCA method used to study single pricing period market is then extended to multiple pricing period market analysis. It is found that the market outcome of a GenCo using a constant supply function across multiple pricing period is contrasted with the case that GenCos using a specified supply function in each pricing period. It is important to observe outcomes of different markets in which obligations for consistent bidding are set up or not.

Owing to the recent California electricity crisis, more and more researchers are convinced that forward market plays an important role for market power mitigated in electricity markets. In this thesis, the issue of whether rational GenCos would voluntarily enter forward markets or not is examined and the factors which could affect the bidding behavior are studied. The thesis formulates a two-settlement electricity market as a two-stage game. The LSFE model and Cournot model are used to model strategic bidding for the spot market, while the forward market is modeled by the Cournot model. GenCos' bidding behaviors are analyzed by CCA and two numerical examples are used to verify the theoretical analysis.

From the works undertaken in this thesis, it is found that the CCA method is robust and flexible and has the potential to be used to solve the complicated equilibrium problems in real-world electricity markets.

ACKNOWLEDGEMENTS

Through out my doctoral studies in the Hong Kong Polytechnic University, many teachers, students and friends have given me generous help and have been contributing to the success of this Ph.D. thesis. I will give my sincere appreciation to them.

Firstly, I would like to express my sincere gratitude to my chief supervisor, Prof. K. P. Wong for his guidance, encouragement and support during my Ph.D. study. Many thanks go to him for his warmest care and greatest encouragement. I would like to extend my gratitude and appreciation to my co-supervisor, Dr. C. Y. Chung for his guidance on high quality research work and encouragement too.

Secondly, I also give my sincere thanks to another co-supervisor, Dr H. Chen for providing me with the essential tool and information for me in doing my research. He has assisted me in solving some of the tough problems I encountered throughout the period of my research.

Lastly, I would like to give my gratitude and appreciation to group members of Computational Intelligence Applications Research Laboratory (CIARLab). I am also grateful to Prof. Zhang Shaohua in Shanghai University. They offered a lot of good advice in my project and encouraged me when I encountered difficulty.

I am very grateful to my parents and my families for their love, encouragement and support. They have given me invaluable support in my study and life.

The last but not the least, the financial support provided by the Research Committee of the Hong Kong Polytechnic University through the award of the studentship is also appreciated.

CONTENT

Abstract	
Acknowledg	ementsV
Content	VI
List of Table	esX
List of Figur	vesXI
Chapter 1	INTRODUCTION1
1.1	Electric Power Industry Deregulation1
1.2	Market Power2
1.3	Market Power Assessing and Market Modeling4
	1.3.1 Market Power Assessing Index
	1.3.2 Analyzing Electricity Market by Economic Models
1.4	Market Analysis with Co-evolutionary Computation16
1.5	Thesis Layout
1.6	List of Publications
Chapter 2	COEVOLUTIOANRY COMPUTATION23
2.1	Introduction
2.2	Evolutionary Computation
	2.2.1 Genetic Algorithm25
	2.2.2 Evolutionary Strategies and Evolutionary Programming27

	2.2.3	Particle Sv	warm Optimiza	tion		•••••		28
2.3	Co-eve	olutionary C	omputation					29
	2.3.1	Developm	ent of Co-evolu	utionar	y Comput	ation		30
	2.3.2	General	Framework	of	Coopera	tive	Co-evolu	tionary
		Computati	ion					31
2.4	Detern	nination of F	Power Market E	quilib	rium Using	g CCA		35
	2.4.1	General F	ramework of El	ectrici	ty Pool M	arket		36
	2.4.2	Determina	tion of Market	Equili	brium Usi	ng CC.	A	37
Chapter 3	DF	TERMINA	ATION AND	AN	ALYSIS	OF	ELECT	ICITY
	MA	ARKET F	EQUILIBIUM	WI	TH TRA	NSIN	IISSION	AND
	CA	PACITY C	CONSTRAINT	'S		•••••	•••••	41
3.1	Introdu	uction						41
3.2	Model	Formulation	n					42
	3.2.1	Market M	odel					43
	3.2.2	Determina	tion of Market	Equili	brium Usi	ng CG	A	46
3.3	Case S	tudies						48
	3.3.1	2-Bus Tes	t System					48
	3.3.2	IEEE 30-E	Bus Test System	1				60
3.4	Conclu	ision						64
Chapter 4	Μ	ULTI-PERI	OD MARKET	SIM	ULATIO	N ANE) ANALY	SIS.67
4.1	Introdu	iction						67
4.2	Marke	t Model For	mulation					69

	4.2.1	Single pricing period market model	72
	4.2.2	Multiple pricing period market model	73
	4.2.3	Market equilibrium with inter-temporal constraints	74
4.3	Co-eve	olutionary Approach to Analyzing Market Equilibrium	75
4.4	Case S	Study	79
4.5	Conclu	usion	87
Chapter 5	AN	NALYZING TWO-SETTLEMENT ELECTRICITY MAR	КЕТ
	EQ	QUILIBRIUM	89
5.1	Introdu	uction	89
5.2	Marke	t Model Formulation	91
	5.2.1	Market assumptions	91
	5.2.2	Equilibrium model for strategic bidding in the spot market	92
	5.2.3	Equilibrium model for strategic contracting in the forward m	arket
			94
5.3	Co-eve	olutionary Computation Approach to Analyzing Market Model.	99
5.4	Case S	Study	102
	5.4.1	Three GenCos study case	102
	5.4.2	Five GenCos study case	108
5.5	Conclu	usion	111
Chapter 6	CO	ONCUSION AND FUTURE WORK	113
6.1	Conclu	usion	113
6.2	Future	Work	115

Appendix					
References	123				

LIST OF TABLES

Table 3.1: Parameters of CGA 48
Table 3.2: Coefficients of demand function and generators' cost function
Table 3.3: Case study results of 2-bus test system
Table 3.4: Coefficients of generators' cost function
Table 3.5: Case study results of IEEE 30-bus test system 63
Table 4.1: CGA parameters79
Table 4.2: Load demands for 24 hours 79
Table 4.3: Coefficients of GenCos 80
Table 4.4: Simulation results of low demand cases 83
Table 4.5: Simulation results of high demand cases 85
Table 5.1: CGA parameters
Table 5.2: Cost coefficients of the three GenCos 102
Table 5.3: Simulation results for three GenCos example 105
Table 5.4: Simulation results for three GenCos example with different slopes of the
demand function106
Table 5.5: Simulation results for three GenCos example with different cost coefficients
Table 5.6: Simulation results for three GenCos example with capacity constraints108
Table 5.7: Cost coefficients of the five GenCos 109
Table 5.8: Simulation results for five GenCos example 109

LIST OF FIGURES

Figure 2.1: Pseudo-code of EA	24
Figure 2.2: Flowchart of GA	27
Figure 2.3: General framework of CCA	32
Figure 2.4: Pseudo-code of parallel CCA	34
Figure 2.5: Electricity market modeling by equilibrium model	36
Figure 3.1: 2-bus test system	48
Figure 3.2: Best response curves for $\overline{F} = 180$ MW	50
Figure 3.3: Evolution process of bidding variables for \overline{F} =180MW	50
Figure 3.4: Best response curves for $\overline{F} = 80$ MW	51
Figure 3.5: Evolution process of bidding variables for \overline{F} =80MW	51
Figure 3.6: Convergence process for $\overline{F} = 180$ MW	53
Figure 3.7: Convergence process for $\overline{F} = 80$ MW	53
Figure 3.8: Best response curves for $\overline{F} = 180$ MW and $\overline{P_{G2}} = 50$ MW	55
Figure 3.9: Evolution process of bidding variables for \overline{F} =180MW and $\overline{P_{G2}}$ =50	0 MW .
	56
Figure 3.10: Evolution process of output for \overline{F} =180MW and $\overline{P_{G2}}$ =50MW	56
Figure 4.1: Flow chart of the CGA	78
Figure 4.2: Market price of lower demand cases	86
Figure 4.3: Market price of lower demand case with inter-temporal constraints	86
Figure 4.4: Market price of higher demand cases	87

CHAPTER 1 INTRODUCTION

1.1 Electric Power Industry Deregulation

Power industry deregulation has taken place in many countries in the past twenty years. The earliest deregulation took place in Chile in the late 1970s. Argentina improved the Chilean model by imposing strict limits on market concentration and by improving the structure of payments. Then, the World Bank was active in introducing electricity markets in other Latin American nations, including Peru, Brazil and Colombia, during the 1990s. A key event for electricity markets occurred in 1990 when UK privatized the UK electricity supply industry. The process was used as a model for the deregulation of several other commonwealth countries, notably Australia and New Zealand.

In different deregulation processes, the institutions and market designs were often very different but underlying concepts were almost the same. These concepts are as follows: separate the contestable functions of generation and retail from the natural monopoly functions of transmission and distribution; establish a wholesale electricity market and a retail electricity market. The role of the wholesale market is to allow trading between generators, retailers and other financial intermediaries both for short-term delivery of electricity and for future delivery periods. A retail electricity market exists when end-use customers can choose their supplier from competing electricity retailers.

In fact, the wholesale and retail markets have been mixed (i.e. hybrid model) in many countries. Many regional markets have achieved some success and the ongoing trend continues to be towards deregulation and introduction of competition. However in 2000/2001 major failures, such as the California electricity crisis and the Enron debacle, caused a slow down in the pace of change and in some regions an increase in market regulation and reduction in competition. However, this trend is widely regarded as a temporary one against the longer term one towards more open and competitive markets.

1.2 Market Power

In this new environment, electricity is traded the same way as other commodities. However, a perfectly competitive market does not exist in practice. Many issues can contribute to the market inefficiency, such as market design flaws, market power, and inherent engineering features of power system operations [1]. Some characteristics of electricity markets facilitate the exercise of market power. These characteristics include inelastic demand, limited transmission capacity, and the requirement that supply and demand must balance continuously. Therefore, market power analysis has received attention in both theory and practice [2, 3]. It is important for market designers to fix flaws in market structure and for GenCos to be able to evaluate their positions and conduct their business. Market power is the ability of a market participant to influence market characteristics (e.g. prices, market share, etc.) to advance its profitability. When applied to a liberalized electricity market, market power would mean the ability of a power company to raise prices and profit above the competitive market price level.

In the electricity supply business, market power normally exists to some extent irrespective of the market structure owing to the following reasons:

- 1. Electricity is an essential commodity of which reliable supply is depended upon for everyday living and business operation. This makes the short-term demand for electricity quite inelastic in response to price changes. Suppliers holding sizable market share can take advantage of this inelasticity to vary prices, where not regulated, without suffering from reduced sales volume and revenue.
- In a competitive market where the price for electricity is largely determined by the supply/demand margin, suppliers having sizable market share can influence the short-term supply/demand margin to drive market price upward.
- 3. The electricity supply business is very capital intensive. Potential new participants often find it very difficult to enter an established market. Even when new generators can be built to compete for supplying existing or new customers, it is often impractical and cost prohibitive to replicate the power grid to deliver electricity to the end users. Lack of access to the power grid virtually blocks any potential new participants from competing with the incumbent suppliers that also own the grid.

Regulating market power and minimizing its abuse are therefore key considerations in ensuring proper market performance. Three types of market power are pertinent to the electricity market: the horizontal, vertical and locational market power.

- Horizontal market power is exercised when an entity profitably drives up prices through its control of a single segment, such as electricity generation. An entity would possess horizontal market power when it owns a significant share of the total generating capacity available to the market.
- 2. Vertical market power is exercised when an entity involved in two related segments, such as electricity generation and transmission, uses its dominance in one segment to raise prices and earn extra profits for the overall enterprise or to disadvantage other suppliers.
- 3. Locational market power is the result of the existence of transmission constraint, which limits the ability for a region to access external supply sources. The local electricity suppliers may therefore charge the local customers a higher tariff without rivalry from external sources.

1.3 Market Power Assessing and Market Modeling

Market power is harmful to competition and it is necessary to identify the potential for its abuse, and such findings have important policy implications. In recent years, much research has been done on investigating the potential for market power abuse in electricity markets [4].

1.3.1 Market Power Assessing Index

Several indexes have been used to assess market power recently. The most widely used index for structural assessments of a market, the Herfindahl–Hirschman Index (HHI), is computed by squaring each supplier's market share, then adding the squared shares. Empirical research in electricity markets also sheds light on the level at which market concentration may have a significant adverse impact on the competitiveness. It is generally agreed that an electricity market composed of five equally sized GenCos would be workably competitive [5].

The HHI method has the advantage of specificity but the drawback that it has no supporting theory. Many researchers such as [6] have pointed out that structural measures of market power, such as the HHI, have several shortcomings when applied to electricity markets; the most obvious is that the demand elasticity is not represented in the HHI formulation. Certainly, the market power of a supplier will change over time as demand conditions change. When the demand increases, the market power of the suppliers will increase. The HHI also fails to consider a supplier's true competitive positions, since consideration of the relative costs of different generation resources or specific geographic factors is absent from the HHI [5].

The Lerner Index (LI) and Price–Cost Margin Index (PCMI) are other two retrospective indicators of market power [7, 8]. The former is defined as LI = (p-MC)/pand the latter is defined as $PCMI = (p-p_c)/p_c$. Here, p, p_c and MC represent actual market price, perfectly competitive price and Marginal Cost (MC) respectively. Since it is hard to obtain MC information in a real market, these two indices do not have much practical value.

In fact, the market power depends not just on market concentration, but also on how demand varies relative to the degree of excess capacity. So another index, a forecast load and total available supply ratio (demand–supply ratio), is proposed in [7], and the New England market data is used to demonstrate it.

1.3.2 Analyzing Electricity Market by Economic Models

The following methods have been developed for market power analysis [4]: analysis of market shares and market concentration; estimation of pricing behavior; modeling market by economic models.

Most researchers focus on market power analysis by using economic models. Researchers have developed three directory branches for market modeling [9]: optimization models, equilibrium models and simulation models. Optimization models focus on the profit maximization problem for one of the GenCos competing in the market, while equilibrium models represent the overall market behavior taking into consideration competition among all participants. Equilibrium and simulation-based models represent market behavior considering competition among all participants. On the contrary, optimization models only represent one GenCo. Consequently, in the latter models, the market is synthesized in the representation of the price clearing process, which can be modeled as exogenous to the optimization program or as dependent of the quantity supplied by the GenCo of interest. The equilibrium models are more suitable to long-term planning and market power analysis since they consider all participants. Simulation models are an alternative to equilibrium models when the problem under consideration is too complex to be addressed within a formal equilibrium framework. This thesis will focus on market analysis with equilibrium models.

A Single-firm optimization models:

The approaches based on the profit maximization problem of one firm (i.e. GenCo) are grouped together into the single-firm optimization category. These models take into account relevant operational constraints of the generation system owned by the firm of interest as well as the price clearing process. For example, the better computational tractability of optimization models enables them to deal with difficult and detailed problems, such as building daily bid curves in the short-term. For instance, Rajamaran *et al.* [10] describe and solve the self commitment problem of a generation firm in the presence of exogenous price uncertainty. A new framework to build bidding strategies for power suppliers in an electricity market is presented with imperfection of knowledge of rival suppliers in [11].

B Equilibrium models:

The oligopoly market equilibrium analysis has been widely used in recent years. In economics, market equilibrium refers to a condition where a market price is established through competition such that the amount of goods or services sought by buyers is equal to the amount of goods or services produced by sellers. This price is often called the equilibrium price or market clearing price and will tend not to change unless demand or supply change. There are several oligopoly equilibrium models available, the most prominent being the Cournot model, Bertrand model, Stackelberg model, Supply Function Equilibrium (SFE) model and conjecture variables model. These models explicitly model the strategic behavior of suppliers. In these models, suppliers decide their bidding strategies and believe that rivals will not alter their strategies, and price is then determined solely by the strategies of the suppliers and the demand curve. Approaches which explicitly consider market equilibria are grouped together into the equilibrium models category. Although these approaches differ in regard to the strategic variables, they are based on the concept of Nash Equilibrium (NE)—the market reaches equilibrium when each firm's strategy is the best response to the strategies actually employed by its opponents.

1. Betrand model

The Betrand oligopoly model assumes that GenCos use the price as the strategy: each strategic GenCo decides its price to produce, while treating the output level of its competitors as a linear function of the price. The Bertrand model is used to predict the electricity prices in [12]. Bertrand model is based on the assumption that any GenCos can capture the entire market by pricing below other suppliers and supplying the entire demand. But generation capacity constraints and increasing marginal costs make this assumption invalid.

2. Stackelberg model

The Stackelberg model is a strategic game in economics in which the leader GenCo moves first and then the follower GenCos move sequentially. In game theory terms, the players of this game are leader(s) and follower(s). Stackelberg model for simulating deregulated electricity markets is introduced in [13]. The model is consisting of one or a few large producers and a larger number of fringe producers. It is assumed that the large producer(s) would adopt oligopoly strategy using their market power while the small producers would use Betrand-like strategy. The electricity is formulated by a Stackelberg model and a noninterior point algorithm is used to solve this Stackelberg equilibrium [14]. Numerical examples illustrate that the leader participant can abuse market power and increase in the presence of tighter transmission capacity constraints.

3. Cournot model

The Cournot oligopoly model assumes that strategic GenCos employ quantity strategies: each strategic GenCo decides its quantity to produce, while treating the output level of its competitors as a constant. The Cournot game is used to model electricity market in [15, 16] and the market models developed exploits the standard approach to interpreting market equilibrium as defining the first order conditions for a related optimization problem. The California electricity market is simulated by a Cournot market model [2]. Simulation results show that there is high market price in high demand periods. Genetic Algorithm (GA) with a new mutation operator is presented and applied to determine the Cournot Game electricity market in [17]. The equilibrium solution is formulated as a single objective GA and accurate and reliable numerical solutions can be obtained. A new approach to find NE in electricity markets is presented in [18]. It is based on the Nikaido-Isoda function and a relaxation algorithm. The method can be seen either as centralized optimization or as distributed optimization, where the generating companies solve their own profit maximization subproblems. The Cournot equilibrium in a transmission-constrained network is investigated in [19]. Simulation results show that pure strategy equilibrium can break down even when a transmission constraint exceeds the value of the unconstrained Cournot equilibrium line flow. The Cournot bidding electricity market with congestion management is modeled as a three level optimization problem in [20]. A statistical methodology is then proposed to solve this problem. A pay off matrix approach is proposed to solve the transmission constrained market equilibrium in [21, 22]. The proposed method obtains the mixed strategy NE that other iterated NE search method is difficult to obtain. The Cournot model can also be used to study the two-settlement electricity markets. Determination of the equilibrium in the markets is modeled as an Equilibrium Problem with Equilibrium Constraints (EPEC) by Cournot model, in which each GenCo solves a Mathematical Program with Equilibrium Constraints (MPEC) in [23]. Cournot model is well tractable and is widely used in market modeling. But it is difficult to handle a market with inelastic demand. However, the short-run demand elasticity in electricity markets is almost zero. As a result, price predictions from Cournot models depend on assumptions about a competitive fringe are not reliable.

4. SFE model

The general SFE model is introduced by Klemperer and Meyer [24]. The generators compete for the spot market by submitting their bids in the form of supply functions, which state the amount they would be willing to produce at any price. Several studies have used SFE models to consider the England and Wales market and other electricity markets. Green and Newbery [25] assume that each GenCo submits a smooth supply schedule, relating amount supplied to marginal price and look for the noncooperative NE of the spot market, which implies a markup on marginal cost. Green [26] further models the effect of three policies that could increase the amount of competition in the electricity spot market in England and Wales. The linear supply function model with asymmetric GenCos is introduced in the analysis. He also analyzes the electricity contract market in England and Wales in [27]. He models competition in the spot market with supply functions and linear marginal costs. Rudkevich et al. [28] model a pool-based electricity market as a noncooperative game with a number of identical profit-maximizing GenCos. They use the analytical solution of SFE and relax the convexity and differentiability conditions to allow for the realistic step-wise supply curves to be studied. Rudkevich further presents a stylized model of the learning process through which GenCos can adjust their supply bidding strategies in order to achieve a rational profit-maximizing equilibrium behavior in the form of SFE [29]. Baldick and Hogan [30] consider a supply function model of an electricity market where strategic GenCos have capacity constraints. They show that if GenCos have heterogeneous cost functions and capacity constraints, then the differential equation approach to finding the equilibrium supply function may not be effective because it produces supply functions that fail to be nondecreasing. They analyze the nondecreasing constraints and characterize piece-wise continuously differentiable equilibrium. To find stable equilibria, they numerically solve for the equilibrium by iterating in the function space of allowable supply functions. Baldick et al. [31, 32] consider an SFE model of interaction in an electricity market, assuming a linear demand function and considering a competitive fringe and several strategic players with

capacity limits and affine marginal costs. They assume that bid rules allow affine or piecewise affine nondecreasing supply function by GenCos and extend the results of Green and Rudkevich concerning the linear SFE solution. The capacity constraints are introduced, and an ad hoc approach to constructing piece-wise affine supply curves is proposed. A new uniform framework of electricity market analysis based on co-evolutionary computation is developed for SFE oligopoly electricity markets in [33]. The piece-wise SFE with nonconvex cost function that are difficult to be handled by the analytical approaches are also investigated.

Recently, some studies have been performed on determination of the SFE with transmission constraints. Baldick [34] investigates SFE in bid-based electricity markets with transmission constraints. He has demonstrated that the parameterization of the supply function model has a significant effect on the calculated results. The market equilibrium is modeled as a two-level optimization problem in which participants try to maximize their profit under the constraints, and their dispatch and price are determined by the Optimal Power Flow (OPF) in [35]. References [36, 37] formulate the problem of calculating SFE in the presence of transmission constraints as a MPEC.

According to these reviews, the SFE, which is originally developed as a way of modeling how competitors could achieve profit maximizing equilibria in the marketplace under conditions of uncertain demand [24], appears to be a promising model of interaction in deregulated power markets. Until now, short-run electricity demand elasticity appears to have been very low, again making the SFE method more applicable to electricity market modeling. SFE has been chosen as the basis for many

12

power market models, at least for those markets where transmission network constraints could be ignored. SFE competes with the Cournot model as a practical tool for studying oligopoly in the electricity industry. SFE is attractive compared to Cournot because it offers a more realistic view of electricity markets, where bid rules may require suppliers to offer a price schedule that may apply throughout a day, rather than simply put forth a series of quantity bids over a day.

5. Equilibrium model with conjecture variable

The Conjectural Variation (CV) method can be used to simulate the strategic behavior of a game with imperfect information in actual electricity market. Another advantage of CV method is that it can easily model a market with different players: leaders, followers, or price takers. The approach of conjectural supply function is proposed by Garcia-Alcalde to simulate the Spain electricity market effectively in [38] and applied by Day and Hobbs in simulation of England and Wales market with linear Direct Current (DC) network [39]. Both papers [38, 39] only concentrate on the static equilibrium, but ignore the dynamic characteristics of CV, which is suitable to be applied in the learning process in the electricity market. A conjectural variation based learning method is proposed for GenCos to improve their strategic bidding performance in a spot electricity market taking account of the expected reaction of their rivals [40]. With the application of conjecture, each GenCo can make its optimal generation decision in the learning process according to available information published in the electricity market. Examples are used to illustrate that motivation is existed for each GenCo to start learning, and learning of all GenCos will decrease the market clearing price of electricity and improve the total social welfare. Conjecture Variable based Bidding Strategy (CVBS) method is used by GenCos to improve their strategic bidding and maximize their profits in real electricity spot markets with imperfect information [41]. A GenCo using CVBS integrates its rivals into one fictitious competitor and estimate its generation and reaction to the GenCo's change of output so that an optimal decision can be made accordingly. It is shown that classical Game Theoretic Bidding Strategies (GTBS) are special cases of CVBS families, and the system equilibrium reached via CVBS is NE. The analytic conclusions have been validated by computer simulations.

C Simulation models:

Simulation models are an alternative to equilibrium models when the problem under consideration is too complex to be addressed within a formal equilibrium framework. Simulation models typically represent each agent's strategic decision dynamics by a set of sequential rules that can range from scheduling generation units to constructing offer curves that include a reaction to previous offers submitted by competitors. The great advantage of a simulation approach lies in the flexibility it provides to implement almost any kind of strategic behavior. However, this freedom also requires that the assumptions embedded in the simulation be theoretically justified.

In many cases, simulation models are closely related to one of the families of equilibrium models. For example, when in a simulation model GenCos are assumed to take their decisions in the form of quantities, the authors will typically refer to the Cournot equilibrium model in order to support the adequacy of their approaches. Otero-Novas *et al.* present a simulation model that considers the profit maximization objective of each GenCo while accounting for the technical constraints that affect thermal and hydro generating units [42]. Day and Bunn [43] propose a simulation model, which constructs optimal supply functions, to analyze the potential for market power in the England & Wales Pool. The modeling flexibility of simulation models allows for a wide range of purposes although there is still some controversy as to the appropriate uses of agent-based models. Bower and Bunn [44] present an agent-based simulation model in which GenCos are represented as autonomous adaptive agents that participate in a repetitive daily market and search for strategies that maximize their profit based on the results obtained in the previous session. The dynamics in two-settlement electricity markets is studied by an agent-based model in [45]. Numerical simulations imply that the access to the forward market leads to more competitive behaviors of the suppliers in the spot market, and thus to lower spot energy prices.

This section presents a comprehensive review of market power related issues in emerging electricity markets, with special emphasis on methods about equilibrium model for analysis of market power. Among these models, the Cournot and SFE models are the most extensively used models for analyzing pool-based electricity markets. Fortunately, direct estimation of the Cournot and SFE equilibria appears to be more feasible for electricity markets than the case for other commodities.

1.4 Market Analysis with Co-evolutionary Computation

Market equilibrium analysis is of fundamental importance because it provides the regulator with relevant information to identify and mitigate the exercise of market power [4]. It also provides GenCos with the appropriate information to maximize their respective profits, within the regulatory framework, by altering market clearing prices to their own respective benefits.

There are many methods, such as empirical analysis [19, 46, 47], agent-based simulation [45, 48-51], iterative NE search algorithms [23, 29, 35, 36, 52] and complementarity program method [37, 53, 54], employed to investigate the market equilibrium.

Although a lot of techniques have been employed to study the market equilibrium, no method has been extensively accepted, and the existing techniques need to be complementary each other. So an alternative tool should be developed to handle these problems. Recently, the application of co-evolutionary computation techniques to electricity market analysis has become a topical research area. Co-evolutionary computation approach has been successfully used to economical simulation and to determine the market equilibrium in complex games [33, 55, 56]. Reference [57] develops a co-evolutionary GA to model industrial organization games and has illustrated the potential of co-evolutionary computation in market simulation. Investigation on the dynamic behavior of market participants using a co-evolutionary approach has been performed in [58]. A hybrid co-evolutionary programming approach for NE search in games with local optima is proposed in [55] and the

16

transmission-constrained electricity market is also studied. A new uniform framework of electricity market analysis based on co-evolutionary computation is developed for analyzing both Cournot [56] and SFE [33] electricity markets. The market players' repeated bidding behavior, nonlinear market models and piece-wise SFE with nonconvex cost function that are difficult to be handled by the analytical approaches are also investigated in [33, 56] by co-evolutionary computation approach. Simulation results have verified that the computation of co-evolutionary computation is highly efficient and is potentially practical.

This thesis will focus on several important issues as follows:

Firstly, when the transmission is considered, the profit function of market players may be nonconcave and discontinuous with many local optima. The conditions for the existence of the market equilibrium in a transmission-constrained power market are very complex. The thesis should reinvestigate the bidding behavior of market players in a constrained market with a powerful tool.

Secondly, the LSFE models with single parameter are investigated in [35-37, 46, 59], but the LSFE with multiple parameters should also be studied. When the capacity constraints are considered in the market analysis, determination of market equilibrium becomes a more challenging task.

Thirdly, when compared to actual markets, some of these works study single pricing period market equilibrium only and most of these works have not considered the inter-temporal constrains of the real market. It will be significative to model the effect of the actual GenCos on market-clearing prices in multi-period market setting with consideration of inter-temporal constraints.

Fourthly, due to the California electricity crisis, more and more researchers are convinced that forward market arrangement plays an important role as a means for market power mitigation in electricity markets [3]. Therefore, market equilibrium models are required to model the competition of generators in spot and contract markets. Then a new methodology need be developed to handle this challenging task.

Finally, co-evolutionary computation is a fast developing research area of evolutionary computation. In particular, co-evolutionary computation is of special importance in both its empirical evidence in market simulation and its underlying theoretical foundations because the dynamic behaviour of co-evolutionary computation is still unclear and need be studied.

This thesis establishes the mathematical models to study the above issues for electricity markets and uses co-evolutionary computation and game theories to analyze the equilibrium of electricity markets.

1.5 Thesis Layout

The organization of the thesis is as follows:

Chapter 1 presents the literature review on power market model and market power assessment and highlights the importance of the research in this thesis.

Chapter 2 provides an introduction to evolutionary computation and co-evolutionary computation. The general market model and market equilibrium are also introduced.

Determination of market equilibrium with a co-evolutionary computation algorithm is then preliminarily presented.

Chapter 3 further analyzes the performance of co-evolutionary computation in solving power market equilibrium problem with capacity constraints and transmission constraints. The effects of transmission and generation capacity constraints on the single pricing period NE in a day-ahead market is investigated. The LSFE and the Cournot market models are employed in the evaluation. The evaluation is based on a parallel Co-evolutionary Computation Algorithm (CCA). The existence or non-existence of the NE due to the above constraints are illustrated using a 2-bus test system and the IEEE 30-bus system. The effects of the different parameters settings of the LSFE model on the equilibria are also studied and compared with those found based on the Cournot model.

Chapter 4 is used to simulate the electricity market in a day-ahead pool-based market setting with co-evolutionary computation. It extends the co-evolutionary computation to analyze market power and bidding strategies in multi-period market with inter-temporal constraints. It reproduces actual market functioning to make the simulation results comparable to actual market settings. It is important to study the outcomes of markets in which the GenCos are compelled to bid consistent for multiple pricing period.

Chapter 5 presents a two-stage game model to formulate the competition of GenCos in bid-based pool spot markets and forward contract markets, as well as the interaction between these two markets. The issue of whether rational GenCos would voluntarily enter contract markets is examined and the factors which could affect the contractual

19

behavior are studied. The effects of GenCos' cost parameters and demand elasticity on the bidding behavior of GenCos are studied. Two numerical examples are used to verify the theoretical results.

Chapter 6 summarizes the finding of the thesis and discusses some potential future works related to equilibrium analysis with the co-evolutionary computation.

1.6 List of Publications

Accepted Journal Papers:

S.X. Zhang, C.Y. Chung, K.P. Wong, H. Chen, "Analyzing Two-settlement Electricity Market Equilibrium by Coevolutionary Computation Approach," Accepted by IEEE transactions on power systems, 2009.

Referred Conference Papers:

S.X. Zhang, C.Y. Chung, K.P. Wong and H. Chen, "Determination of NE in Transmission-constrained Power market Using Nash Genetic Algorithm", International Conference on Electrical Engineering (ICEE) 2006, July, 2006, Korea.

Submitted Journal Papers:

S.X. Zhang, C.Y. Chung, K.P. Wong and H. Chen, "Equilibrium Analysis of Transmission and Generation Capacity Constrained Electricity Market Using Cooperative Coevolutionary Computation", submitted to Operations Research.

Journal Paper under Preparation:

S.X. Zhang, C.Y. Chung, K.P. Wong and H. Chen, "Multi-period Market Simulation: Using a Coevolutionary Computation Approach,"

CHAPTER 2 COEVOLUTIOANRY COMPUTATION

2.1 Introduction

Evolutionary Algorithms (EA) in the field of Evolutionary Computation (EC) have been widely applied in solving optimization problems, which are previously difficult or impossible to be solved. These algorithms include GA, Evolution Strategies (ES), Evolutionary Programming (EP) and Particle Swarm Optimization (PSO) etc. Recently, in order to solve extremely challenging problems, EAs have been combined among themselves and with knowledge elements of the problems, as well as with traditional approaches. EAs may offer many advantages such as shorter time for program development, strong solution searching capability for nonlinear and discrete problems, and robust performance with relatively insensitive to noisy and/or missing data.

This chapter will first provide an overview of EC and introduce the basic theory of EC and co-evolutionary computation; and then apply the co-evolutionary computation to determine the market equilibrium and analyze the power market.
2.2 Evolutionary Computation

There are a number of different classes of algorithms that make up the field of EC. They are notated as GA, EP, ES and Genetic Programming (GP). Although these four classes of EAs are different, they are all based on the same fundamental principles of Darwinian evolution. Another kind of evolution computation algorithms is swarm intelligence, such as Ant Colony Optimization (ACO) and PSO. EAs can be loosely recognized by the following criteria: iterative progress, growth or development; population based; guided random search; parallel processing; often biologically inspired. Finally a general framework of EA is given in Figure 2.1.

```
Procedure of EA

t = 0;

Initialize Pop(t);

Evaluate Pop(t);

While (End condition is not satisfied)

{

Parents(t) = Select_Parents(Pop(t));

Offspring(t) = Procreate(Parents(t));

Evaluate(Offspring(t));

Pop(t+1)= Replace(Pop(t), Offspring(t));

t = t + 1;

}
```

Figure 2.1: Pseudo-code of EA

2.2.1 Genetic Algorithm

GA was formally introduced in the 1970s by John Holland at University of Michigan [60]. A GA is a search technique used in computing to find true or approximate solutions to optimization and search problems. GAs are a particular class of EAs that use techniques inspired by evolutionary biology such as inheritance, mutation, selection, and crossover (also called recombination) [60].

GAs are implemented as a computer simulation in which a population of abstract representations (i.e. chromosomes or genotype or genome) of candidate solutions (i.e. individuals or phenotypes) to an optimization problem evolves toward better solutions. The evolution usually starts from a population of randomly generated individuals and happens in generations. In each generation, the fitness of every individual in the population is evaluated. Multiple individuals are stochastically selected from the current population (based on their fitness), and modified (recombined and possibly mutated) to form a new population. The new population is then used in the next iteration of the algorithm.

The flowchart of the classic GA is showed in Figure 2.2 and its procedure is described as follows:

- 1. [Starting] Generate random population of chromosomes (suitable solutions for the problem)
- 2. [Fitness evaluation] Evaluate the fitness of each chromosome in the population
- 3. [Creating new population] Create a new population by repeating following steps until the new population is complete

- [Selection] Select two parent chromosomes from a population according to their fitness (the better fitness, the bigger chance to be selected)
- [Crossover] With a crossover probability, cross over the parents to form new offspring (children). If no crossover is performed, the offspring is the exact copy of parents.
- [Mutation] With a mutation probability, mutate new offspring at each locus (position in chromosome).
- [Accepting] Place new offspring in the new population
- 4. [Replacing] Use new generated population for a further run of the algorithm
- 5. [Testing] If the termination condition is satisfied, stop, and return the best solution in the current population
- 6. [Looping] Go to step 2



Figure 2.2: Flowchart of GA

2.2.2 Evolutionary Strategies and Evolutionary Programming

ES [61] employs real-coded variables and, in its original form, it relies on mutation as the search operator. It has evolved to share many features with GA. The major similarity between these two types of algorithms is that they both maintain populations of potential solutions and use a selection mechanism for choosing the best individuals from the population. The main differences are: GAs rely mainly on recombination to explore the search space, while ES uses mutation as the dominant operator; and ES is an abstraction of evolution at individual behavior level, stressing the behavioral link between an individual and its offspring, while GAs maintain the genetic link.

EP [62] is a stochastic optimization strategy similar to GA, which places emphasis on the behavioral linkage between parents and their offspring, rather than seeking to emulate specific genetic operators as observed in nature. EP is similar to ES, although the two approaches are developed independently. Like both ES and GAs, EP is a useful method of optimization when other techniques such as gradient descent or direct analytical discovery are not possible. Combinatorial and real-valued function optimization in which the optimization surface or fitness landscape is "rugged" are well suited for EP.

2.2.3 Particle Swarm Optimization

In 1995, Kennedy and Eberhart first introduced the PSO method, motivated by social behavior of organisms such as fish schooling and bird flocking [63]. PSO, as an optimization tool, provides a population-based search procedure in which individuals (i.e. particles) change their positions (i.e. states) with time. In a PSO system, particles fly around in a multidimensional search space. During flight, each particle adjusts its position according to its own experience, and the experience of neighboring particles, making use of the best position encountered by itself and its neighbors. The swarm direction of a particle is defined by the set of particles neighboring the particle and its history experience. PSO is an exciting new methodology in evolutionary computation. It is somewhat similar to a GA in that the system is initialized with a population of

random solutions. Unlike other algorithms, each potential solution (i.e. a particle) is also assigned a randomized velocity and then flown through the problem hyperspace. PSO has been found to be extremely effective in solving a wide range of engineering problems. It is very simple to implement (the algorithm comprises two lines of computer code) and solve problems very quickly [64].

PSO is described as follows:

$$v_{i,j}^{(t+1)} = wv_i^{(t)} + c_1 rand()(pbest_{i,j} - x_{i,j}^{(t)}) + c_2 rand()(gbest_j - x_{i,j}^{(t)})$$

$$x_{i,j}^{(t+1)} = x_{i,j}^{(t)} + v_{i,j}^{(t+1)}$$
(2.1)

where

$i \in 1,,N$	N is the number of population size
$j \in I,,d$	d is the number of dimension
$t \in 1,,T$	T is the number of maximum generation
W	inertia weight factor
<i>c</i> ₁ , <i>c</i> ₂	acceleration constant
$v_{i,j}^{(t)}$	uniform random value in the range [0,1]
$x_{i,j}^{(t)}$	velocity of particle at iteration
$pbest_{i,j}$	the best previous position of the i th particle
gbest ;	the value of $pbest_i$ with the lowest fitness

2.3 Co-evolutionary Computation

The simplest definition about a co-evolutionary algorithm is that "a co-evolutionary algorithm is an EA (or collection of EAs) in which the fitness of an individual depends on the relationship between that individual and other individuals" [65]. Such a definition immediately imbues these algorithms with a variety of views differing from those of standard EAs. For example, one might favor the view that individuals are not evaluated at all, but in fact their interactions are evaluated. Alternatively, one might

look at individual fitness evaluation from the perspective of a dynamic landscape, given that the result of the evaluation is contextually dependent on the state of other individuals. In either case it is clear that they differ in profound ways from the traditional EAs.

2.3.1 Development of Co-evolutionary Computation

In fact, most works in co-evolutionary algorithms are in the area of competitive coevolution in the early time. The most popular competitive coevolution has been applied to game playing strategies [66]. Competition has played a vital part in attempts to coevolve complex agent behaviors [67] so coevoluiontary algorithm has also been applied to a variety of machine learning problems.

Potter and De Jong first propose cooperative evolutionary computation by developing a cooperative co-evolutionary algorithm for static function optimization through problem decomposition [68]. In this model, each population contains individuals representing a component of a larger solution. The populations evolve almost independently and interact only to obtain their fitness. Fitness of an individual representing a subcomponent is equal to the quality of the whole-problem solution obtained by assembling that individual with individuals representing the other subcomponents (for cooperative co-evolutionary computation, these individuals are usually called collaborators). Wiegand [69] attempts to make the algorithm more adaptively allocate resources by allowing migrations of individuals from one population to another. Since initial results of this method are promising, a framework for using cooperative co-evolutionary computation algorithms is developed in [70] and further extended in [65].

The main difference between cooperative and competitive co-evolutionary algorithms is the way that their individuals interact with each other. In the case of cooperative algorithms, individuals are rewarded when they work well with other individuals and punished when they perform poorly together. In the case of competitive algorithms, however, individuals are rewarded at the expense of those with which they interact. In this thesis, the Cooperative Co-evolutionary Algorithm (CCA) is employed.

CCA has promising applications in power system optimization and power market simulation. The decomposition and coordination methods based on the CCA are used in unit commitment [71] and reactive power optimization [72]. Another recent fast-developing area is the application of CCA [73] in determining the game equilibrium or analyzing the oligopolistic market [33, 55, 56]. CCA is found to be an effective and powerful approach for oligopolistic market analysis.

2.3.2 General Framework of Cooperative Co-evolutionary Computation

The basic framework of CCA is described in this section. It simulates the co-evolutionary mechanism in nature and adopts the notion of ecosystem. Multiple species coevolve and interact with each other and result in the evolution of the ecosystem. Each species evolves a bundle of individuals through the repeated application of a conventional EA. The individuals in the species are genetically isolated. But one species can interact with the others within a shared domain model.

Figure 2.3 shows the fitness evaluation phase of the EA from the perspective of species i. To evaluate an individual from species i, collaborators are formed with representatives from each of the other species. The domain model solves for the system variable. Then species i can use the system variable to evaluate the fitness of its individual. There are many possible methods for choosing representatives (i.e. collaborators). An obvious one is to simply let the current best individual from each species be the representative, and an alternative one is to randomly select an individual from each species to be the representative [65]. Here, the representative is the best individual in a species and is used as the collaborator.



Figure 2.3: General framework of CCA

Wigand points out the many parameters that need to be set in CCA [65], such as which method of EA is used in each population and how the populations interact. CCA has a set of coevolution specific parameters whose values can greatly affect performance. It would be useful to have some knowledge of the effects of the reasonable values of parameters, in order to select a better one. In recent years, a lot of effort has been spent on generating such knowledge for CCA. Most studies [74-77] focus on collaboration schemes such as how many individuals should be used for evaluation, how these individuals should be selected, and how the outcomes of their interactions are aggregated. An initial study on the performance effects of population size and elitism is performed in [78] and then extended in [79]. But most of these studies [74-79] focus on using the decomposition technology to study a single objective optimization problem. The knowledge and methods of CCA parameters setting may not be suitable for the application of CCA in the game equilibrium determination. Indeed, this thesis has found that the CCA is observed to be more effective and efficient for market equilibrium determination when selection with elitism is used, the best individual is selected as representative, rather large mutation probability and population size are used.

A parameter called update timing which controls whether the CCA runs its species sequentially or in parallel [65] is introduced in this section. The term "sequential" refers to processing each population in a sequential order and choosing representatives from the current state of the other populations. Since the populations can change during each round, the order of the population selected for processing will affect the results. An alternate approach is the parallel one in which the individual populations are processed in parallel and synchronized only at the end of each round. In this approach, representatives are selected from the previous generation from each population. These two approaches are studied in [80] for pseudo-boolean functions and in [81] for functions defined on continuous real-number domains. Studies show the performance of the method is dependent on the problem property.

The pseudo-code of a CCA with parallel framework is given in Figure 2.4. The evolution of each species (i.e. population) is handled by an EA. A species means a population of EA in this algorithm and the species coevolves in parallel.

```
g = 0
for i=1:Num of Species
     initialize the species population Pop_s^s in Species i
     evaluate fitness of each individual in Pop^{g}
     choose a collaboration Col_i^{g} from Pop_i^{g}
end
while g<max generations
                               do
  for i=1: Num of Species
        begin
           reproduction from Pop_i^k to get Parent<sub>i</sub><sup>g</sup>
           from Parent_i^g to get Offspring_i^g (i.e. Pop_i^{g+1})
          evaluate fitness of each individual in Pop_{s}^{g+1}
  end
  for i=1: Num of Species
        choose a collaboration Col_i^{g+1} from Pop_i^{g+1}
  end
    g = g + 1
end
```

Figure 2.4: Pseudo-code of parallel CCA

In CCA, species can calculate their best strategy based on information about what other species have done in the last generation [79]. Individuals in the species with higher fitness are at a productive advantage compared to individuals with lower fitness; hence the latter decreases in frequency in the new population (natural selection). In this situation species are viewed as being coded with a strategy and selection pressure favors species that are fitter whose strategy yields a higher payoff against the population. Species adopt actions that optimize their expected payoff given what they expect others to do.

Since CCA explicitly models the reciprocity among the coevolving species, it embodies a dynamic process of strategy choice and interaction, which coincides with the framework of game theory. Recently, game theory, especially evolutionary game theory has begun to be used to analyze the dynamic behaviors of co-evolutionary approach [75]. Co-evolutionary computation is a new area in the research of evolutionary computation. Its theory and applications are still rapidly developing [79-83].

2.4 Determination of Power Market Equilibrium Using CCA

In a non-cooperative game, each player optimizes its own profit given that all the strategies of other players are fixed. The game will reach the NE when no player can further improve its profit by changing its strategy unilaterally [84]. For an *I*-player game, each player has its strategy set X_i and its payoff function is $\pi_i(x_1,...,x_i,...,x_I)$ where the strategy $x_i \in X_i$ and i = 1,..., I. Each player can only choose one strategy from its strategy set. If the strategy profile $(x_1,...,x_i,...,x_I)$ is the NE, it must satisfy the following equation:

$$\pi_i(x_1, \dots, x_i^*, \dots, x_N) \le \pi_i(x_1, \dots, x_i, \dots, x_N)$$
(2.2)

where $x_i^* \in X_i$.

2.4.1 General Framework of Electricity Pool Market

A general game equilibrium model of Bid-Based Pool (BBP) electricity market is introduced in this section. The market is modeled by a noncooperative game model, such as Cournot and LSFE models. It is assumed that *I* market players try to maximize their profit functions $f_i(x_1, x_2, ..., x_I)$ where the decision variables $x_i \in X_i$ in the market. X_i is the decision variable spaces (strategy set) of Player *i*. Each player submits its optimal trading strategy to the Independent System Operator (ISO) or Market Operator (MO). The trading strategy is different for different market models. Here for the standard Cournot model with quantity decision participants, the trading strategy is the quantity to be generated. ISO then calculates the market price according to the demand characteristics and market rules. ISO uses the market players' bidding strategies to determine the market signal (i.e. market price and players' transactions). Each player can calculate its profit with the market price and its trading strategy.



Figure 2.5: Electricity market modeling by equilibrium model

2.4.2 Determination of Market Equilibrium Using CCA

Many methods, such as empirical analysis [19, 46, 47], agent-based simulation [45, 48-51], iterative NE search algorithms [23, 29, 35, 36, 52] and complementarity program method [37, 53, 54], have been employed to determine the market equilibrium. Besides, some heuristic methods , such as learning and coordination between different players [85, 86], have been used to study the economic game.

Empirical analyses on market equilibrium are performed in [19, 46, 47]. The equilibrium is obtained by drawing the best response curves of the players, but this methodology is limited to small test systems.

Iterative NE search algorithms that use repeated individual profit maximization are employed to determine the NE in [23, 29, 35, 36]. Rudkevich [29] presents a model of the learning process in which each GenCo starts from the perfectly competitive supply function and adjusts the slope of supply function through a stylized formula and using the market observed data. Rudkevich proves that GenCos can achieve an SFE through this learning process. Iterative NE search algorithms have been applied to more complex games in [35, 36]. Hobbs *et al.* formulate the maximization problem of individual profits for each GenCo in a transmission constrained electricity market as a MPEC [36], and apply a penalty interior point algorithm to solve the problem. Weber and Overbye formulate the problem to determine the market equilibrium in the power market as a two-level optimization problem [35]. A modified Newton step is merged into the algorithm to optimize the individual profit. The iterative NE search algorithms that use repeated individual profit maximization are hard to handle a complex market because they often use local search methods [35, 36] with limited search step to solve the market players' profit maximization problems.

Another way to determine the market equilibrium is using the Linear Complementary Program(LCP) [53] or Nonlinear Complementary Program (NLP) [37]. Stoft applies LCP to a two-player game in an electricity market [53]. Although the algorithms have been well proven mathematically, they are limited to the bimatrix game. Wang *et al* determine the electricity market equilibrium with transmission constraints using the NLP method [37]. But the studies in [37] have not shown how to handle the situation when the market has no pure equilibrium or has an equilibrium continuum. In addition, the solution obtained by the NLP method may not be a pure NE when the optimization problem is not generalized convex nor are its constraints [87].

It is difficult to determine the NE in a complex game; and even impossible if the payoff functions are non-differentiable functions. So the CCA has been recently used as an alternative tool to determine the NE in power market [33, 55, 56].

In the CCA, each player is represented by a species. Each player will optimize its own benefit function and determine the best solution of its decision variable using an EA by fixing the decision variable of other players. To evaluate individuals from one species, collaborators (i.e. the representatives of each species) from each of the other species are needed. The last generation best decision variable of each player for parallel CCA or current best decision variable of each player for sequential CCA will be selected as the representative and sent to another player for next generation. The process will stop when the maximum number of generation is reached. In a multiple players' game, when the CCA converges to a solution at generation k, each player will not change its best variable because it could not find a better solution without changing the decision variable of another player. This means the fitness of species *i*'s individual in generation l(l > k) is no larger than the best fitness of species *i* in generation *k*.

$$fitness_{i}(x_{1best}^{k},...,x_{i}^{l},...,x_{lbest}^{k}) \leq fitness_{i}(x_{1best}^{k},...,x_{ibest}^{k},...,x_{lbest}^{k})$$
(2.3)

where $fitness_i(x_1, ..., x_i, ..., x_I)$ is the fitness function for species *i*, i.e. its payoff function. By comparing (2.2) and (2.3), if species *i* uses the player *i*'s payoff function $\pi_i(x)$ as the fitness function, (2.3) is satisfying the definition of the NE.

GA is found be more suitable for economic learning in [88]. It is shown that economic learning via GAs can be described as a specific form of an evolutionary game. The performances of different EAs, such as GA, EP and PSO, in simulating players' behaviors are compared [89]. The authors have pointed out that EP and PSO can be more appropriate than GA if solution space is well defined and both of them can utilize the previous global information. On the other hand, if the solution space is highly skewed and tortuous, the previous global information may be wrong and mislead the searching procedure in PSO and EP. Thus, GA will outweigh EP and PS in the later case. In a market game, the relationship and interaction of the market players are very complex. The dynamic behavior of one player is affected by strategies of other market players. The fitness landscape of a player's payoff function is usually unpredictable. Thus, GAs will be used to simulate bidding behaviors of GenCos in the CCA in this thesis.

CHAPTER 3 DETERMINATION AND ANALYSIS OF ELECTICITY MARKET EQUILIBIUM WITH TRANSIMISSION AND CAPACITY CONSTRAINTS

3.1 Introduction

Different equilibrium models have been used for the electricity market analysis including Cournot, Betrand, Stackelberg, SFE and Collusion models [4, 13, 90]. SFE and Cournot models are extensively used in market power analysis. There are many successful application cases to simulate the power market using the SFE models [25, 26, 29, 30, 91] and Cournot models [56, 90]. However, most of these studies neglect the effect of the transmission constraints.

Recently, some studies [19, 35-37, 46, 47, 53-55, 59, 87, 92-94] have been performed on determination of the market equilibrium with transmission constraints. A two-level optimization model is used in which participants try to maximize their profits under the transmission constraints, and their power dispatch and price are determined by the OPF.

When the transmission is considered, the profit function of market players may be nonconvex and discontinuous with many local optima. The conditions for the existence of the market equilibrium are very complex. The following three possible situations exist:

- 1. One pure NE exists when the transmission constraints are not so tight.
- 2. No pure NE exists due to the transmission congestion [19, 46, 47].

3. A NE continuum in LSFE market models appears due to transmission congestion [35, 46, 47].

The LSFE models with single supply function parameter are investigated in [35-37, 46, 59], but the LSFE with multiple supply function parameters should also be studied. When the generation capacity constraints are considered in the market analysis, determination of market equilibrium becomes a more challenging task. A new global search method needs to be explored for market analysis and equilibrium determination in more realistic and complicated conditions.

In this chapter, electricity markets with transmission and generation capacity constraints are studied. A parallel Co-evolutionary Genetic Algorithm (CGA) is applied to determine the market equilibrium with transmission and generation capacity constrains. The performance of the parallel CGA is tested and the effects of the transmission constraints and generation capacity constraints on the behavior of market players and on the existence of the NE are investigated.

3.2 Model Formulation

Suppose there are *I* generators, *J* demands and *N* buses in the power system. The *I* generators belong to *F* GenCos. Generator *i* belonging to GenCo *f* is represented by $i \in S_f$. Each generator is assumed to have a strictly concave quadratic cost function:

$$C_{i}(P_{Gin}) = \frac{1}{2}a_{i}P_{Gin}^{2} + b_{i}P_{Gin} + c_{i} \quad i = 1, \dots, I$$
(3.1)

where P_{Gin} is the quantity generated by Generator *i* at bus *n*; a_i, b_i and c_i are the coefficients of the Generator *i*'s cost function; $C_i(P_{Gin})$ is the cost of generator *i*. Therefore the marginal cost function is affine for each generator as follows:

$$\frac{dC_i}{dP_{G_{in}}}(P_{G_{in}}) = a_i P_{G_{in}} + b_i \quad i = 1, \dots, I$$
(3.2)

Assume that each demand has an inverse linear demand function:

$$p_n(P_{Djn}) = -d_j P_{Djn} + e_j \qquad j = 1,...,J$$
 (3.3)

where p_n and P_{Djn} are the price and the quantity of demand *j* at bus *n* respectively; d_j and e_j are the coefficients of the demand function with $d_j \ge 0$, $e_j \ge 0$. Therefore the demand benefit function can be represented by:

$$B_{j}(P_{Djn}) = -\frac{1}{2}d_{j}P_{Djn}^{2} + e_{j}P_{Djn} \qquad j = 1,...,J$$
(3.4)

where $B_i(P_{Din})$ is the cost of demand *j*.

If the transmission loss is assumed to be negligible, the aggregate demand will be equal to the total output of all the generators in the market as shown below:

$$\sum_{i=1}^{I} P_{G_{in}} = \sum_{J=1}^{J} P_{D_{jn}}$$
(3.5)

3.2.1 Market Model

LSFE model and the less competitive market model, Cournot model, are described in this section.

1) LSFE Model

In the day-ahead power market, the generators are assumed to submit a bid to ISO, in the form of Linear Supply Function (LSF) in each pricing period (i.e. one hour). Each LSF includes an intercept and a slope as follows:

$$p_n(P_{G_{in}}) = \alpha_i P_{G_{in}} + \beta_i \quad i = 1, \dots, I$$
(3.6)

where α_i and β_i are the bidding strategies of the Generator *i*.

Baldick summarizes the parametrization of supply function into four categories [34]. The notations are different from [34].

- 1. *a*-parametrization, where Generator *i* can choose α_i in (3.6) arbitrarily but is required to specify a fixed and pre-chosen value of β_i . In this chapter, $\beta_i = b_i$ is selected.
- 2. *b*-parametrization, where Generator *i* can choose β_i in (3.6) arbitrarily but is required to specify a fixed and pre-chosen value of α_i . In this chapter, $\alpha_i = a_i$ is selected.
- 3. $(a \propto b)$ -parametrization, where Generator *i* can choose α_i and β_i subject to the condition that α_i and β_i have a fixed linear relationship. The LSF described in (3.7) is studied in this chapter, where k_{Gi} is the bidding variable:

$$p_n(P_{G_{in}}) = k_{G_i}(a_i P_{G_{in}} + b_i) \quad i = 1, ..., I$$
 (3.7)

4. (a,b)-parametrization, where Generator *i* can choose α_i and β_i arbitrarily.

Then each GenCo will maximize its own profit where the decision variables are α_i and β_i . The optimization problem can be formulated as:

$$Max \quad \pi_f(\alpha_1, \beta_1, ..., \alpha_i, \beta_i, ..., \alpha_I, \beta_I) = \sum_{i \in S_f} (p_n P_{Gin} - C_i(P_{Gin})) \qquad f = 1, ..., F$$

$$(3.8)$$

where p_n , the nodal price at bus *n*, is determined by ISO.

Once each generator given its bidding, the ISO needs to solve the DC OPF problem in (3.9) for determining the corresponding generating output of each generator ($P_{G_{in}}$), the load demand of each demand ($P_{D_{jn}}$) and the nodal price (p_n). The objective of this market clearing problem is to maximize a quasi-social welfare (defined by the benefit of demands minus the submitted costs of the generators), subject to power balance constraint, transmission constraints and generation capacity constraints.

$$Max \quad \Gamma_{ISO} = \sum_{j=1}^{J} B_{j}(P_{Djn}) - \sum_{i=1}^{I} C_{i}(P_{Gin})$$
$$= \sum_{j=1}^{J} (-\frac{1}{2} d_{j} P_{Djn}^{2} + e_{j} P_{Djn})$$
$$- \sum_{i=1}^{I} (\frac{1}{2} \alpha_{i} P_{Gin}^{2} + \beta_{i} P_{Gin})$$

s.t. $\mathbf{H}^{T}(\mathbf{P}_{G} - \mathbf{P}_{D}) = 0$ Power balance constraint (3.9) $\mathbf{T}(\mathbf{P}_{G} - \mathbf{P}_{D}) \leq \overline{\mathbf{F}}$ Transmission constraints $\underline{\mathbf{P}_{G}} \leq \mathbf{P}_{G} \leq \overline{\mathbf{P}_{G}}$ Capacity constraints of generators $\underline{\mathbf{P}_{D}} \leq \mathbf{P}_{D} \leq \overline{\mathbf{P}_{D}}$ Capacity constraints of demands

where matrix \mathbf{T} is the sensitivity of the branch flows to nodal net injections, i.e. shift factors; \mathbf{H} is a properly dimensioned vector with all ones.

The Lagrangian function for this ISO optimization problem is:

$$\max \Gamma_{ISO} = \sum_{j=l}^{J} B_{j}(P_{Djn}) - \sum_{i=l}^{l} C_{i}(P_{Gin}) + \lambda \mathbf{H}^{T}(\mathbf{P}_{G} - \mathbf{P}_{D}) + \boldsymbol{\mu}^{T}(\mathbf{T}(\mathbf{P}_{G} - \mathbf{P}_{D}) - \overline{\mathbf{F}}) + \hat{\boldsymbol{\tau}}^{T}(\mathbf{P}_{G} - \overline{\mathbf{P}_{G}}) + \check{\boldsymbol{\tau}}^{T}(\underline{\mathbf{P}_{G}} - \mathbf{P}_{G}) + \hat{\boldsymbol{\upsilon}}^{T}(\mathbf{P}_{D} - \overline{\mathbf{P}_{D}}) + \check{\boldsymbol{\upsilon}}^{T}(\underline{\mathbf{P}_{D}} - \mathbf{P}_{D})$$
(3.10)

where λ is the Lagrangian multiplier of power balance constraint; μ , $\hat{\tau}$, $\tilde{\nu}$, $\tilde{\upsilon}$, $\tilde{\upsilon}$ are the corresponding Lagrangian multipliers vector associated with transmission constraints, generators' upper and lower generation capacity limits as well as demands' upper and lower demand capacity limits respectively. The nodal price can be obtained as follows:

$$p_n = \lambda + \sum_{m=1}^{M} \mu_m T_{mn}$$
 $n = 1,...,N$ (3.11)

where T_{mn} is the element (m, n) of matrix **T**; and *M* is the number of transmission lines.

2) Cournot Model

In Cournot model, the generators are assumed to use their quantities as bidding variables. The optimization problem of GenCo f is described in (3.12) where the decision variables are P_{Gin} ($i \in S_f$).

$$Max \quad \pi_{f}(P_{G1n},...,P_{Gin},...,P_{GIn}) = \sum_{i \in S_{f}} (p_{n}P_{Gin} - C_{i}(P_{Gin})) \qquad f = 1,...,F$$
(3.12)

Since P_{Gin} of generators are known, ISO can determine the corresponding P_{Djn} and p_n similar to the optimization problem in (3.10) of the LSFE model, but with the following equation:

$$Max \qquad \Gamma_{ISO} = \sum_{j=1}^{J} B_{j} (P_{Djn}) + \lambda \mathbf{H}^{T} (\mathbf{P}_{G} - \mathbf{P}_{D}) + \boldsymbol{\mu}^{T} (\mathbf{T} (\mathbf{P}_{G} - \mathbf{P}_{D}) - \overline{\mathbf{F}}) + \hat{\boldsymbol{\tau}}^{T} (\mathbf{P}_{G} - \overline{\mathbf{P}_{G}}) + \bar{\boldsymbol{\tau}}^{T} (\underline{\mathbf{P}_{G}} - \mathbf{P}_{G}) + \hat{\boldsymbol{\upsilon}}^{T} (\mathbf{P}_{D} - \overline{\mathbf{P}_{D}}) + \bar{\boldsymbol{\upsilon}}^{T} (\underline{\mathbf{P}_{D}} - \mathbf{P}_{D})$$
(3.13)

3.2.2 Determination of Market Equilibrium Using CGA

Equilibrium determination of an electricity market with *F* GenCos (market players) can be considered as a two-level optimization problem, in which the first level is the ISO's market clearing optimization problem in (3.10) for LSFE model or (3.13) for Cournot model used to determine the market signal including p_n , P_{Gin} and P_{Djn} ; and the second level is the maximization of GenCos' benefits in (3.8) for LSFE model or (3.12) for Cournot model used to optimize the corresponding bidding variables of each GenCo [35].

CCA is applied to solve this problem. Each GenCo is represented by a species, which coevolves using the standard GA operators including crossover and mutation [60].

The procedures can be summarized as follows:

Step 1: Set the basic parameters of CGA including maximum generation number, population size of each species, crossover rate and mutation rate.

Step 2: The individual of each species, which includes the bidding variable of the GenCos, is initialized. All the initial representatives of each species are set randomly.

Step 3: The standard GA is applied to solve the optimization problem of each GenCo in (3.8) or (3.12). For each individual in the species, the ISO market clearing problem in (3.7) is first solved by the quadratic programming technique to determine the corresponding p_n and P_{Gin} for the generator or P_{Djn} for the demand; and then this individual is assigned a fitness value according to the profit function in (3.8) or (3.12). The number of generation for the standard GA is set to be the number of decision variables in species *i*.

Step 4: The best individual of each species (i.e. individual with the highest fitness value) is selected as the representative and sent to other GenCos.

Step 5: Repeat Steps 3-4 using the updated collaborators and also keeping the best strategy of each GenCo in the population (i.e. elitism) until the maximum generation number is reached.

3.3 Case Studies

In this section, the proposed method is applied to a 2-bus test system and the IEEE 30-bus test system. Results for perfect competition are also presented in both examples for the purpose of comparison with the LSFE results. The CGA parameters are given in Table 3.1.

Parameters	Description
Variable code	20 bit binary code
Initial population	Randomly initialized population
Population size	60
Maximum generation	200
Mutation	Bit-flip mutation (mutation probability= 0.1)
Crossover	Two-point crossover (crossover probability= 0.9)
Selection	Roulette selection and 5% population is elitism

Table 3.1: Parameters of CGA

3.3.1 2-Bus Test System

The test system with two generators and one demand is shown in Figure 3.1. The coefficients of generator's cost and demand functions are given in Table 3.2 [35]. The two generators belong to two GenCos respectively.



Figure 3.1: 2-bus test system

The LSFE under $(a \propto b)$ -parametrization is used to illustrate the convergence process of the proposed algorithm in the 2-bus electricity market with transmission constraints and generation capacity constraints. The demand constraints is set as $\underline{P_{D2}} = 0$ MW and $\overline{P_{D2}} = e_i/d_i = 375$ MW. The upper and lower limits of k_{Gi} are assumed to be 20 and 0 respectively.

Bus number	Coefficient	1	2
Domand function	$d_j((MW)^2h)$	0	0.08
Demand Tunction	e_j (\$/MWh)	0	30
	$a_i ((MW)^2h)$	0.02	0.02
Generators' cost function	b_i (\$/MWh)	10	10
	<i>C</i> _{<i>i</i>} (\$/h)	0	0

Table 3.2: Coefficients of demand function and generators' cost function

A Effect of Transmission Constraints:

When the transmission limit (\overline{F}) is set to 180MW, which is a relatively large value, the best response curves are determined to identify whether there is a pure NE. For example, given a k_{G1} in each step, k_{G2} is varied within its decision range to find the corresponding value that gives the maximum profit of Generator 2. Repeating the above calculation step by step yields the best response curve of Generator 2. Similarly, the best response curve of Generator 1 can be obtained. The best response curves of both generators are shown in Figure 3.2. The intersection of the best response curves is the pure NE [84]. In this case, a unique NE is obtained and the proposed algorithm converges to the NE rapidly as shown in Figure 3.3.

When the transmission line becomes tight and \overline{F} is reduced to 80MW, no pure NE exists because there is no intersection point of the best response curves as shown in

Figure 3.4. In Figure 3.5, it is also observed that the CGA will not converge to any solution.



Figure 3.2: Best response curves for $\overline{F} = 180$ MW



Figure 3.3: Evolution process of bidding variables for \overline{F} =180MW



Figure 3.4: Best response curves for $\overline{F} = 80$ MW



Figure 3.5: Evolution process of bidding variables for \overline{F} =80MW.

In order to show the convergence process of CGA, the 2-bus test system is studied again by CGA with a large population size. All other CGA parameters in Table 3.1 are unchanged, except for the population size changed to 1000 which is a considerable large number. In each generation, each species will have a large probability to find its best response variable because the species population is very large. In Figures 3.6 and 3.7, the evolution direction of each species in each generation is to the generator's best response variable curve in which the generator will have the maximum profit. For example, the best variable set of the generators is $k^0 = (k_{1best}^0, k_{2best}^0) = (1, 1)$ in generation 0. Generator 1 will be attracted to k_{1best}^1 . Then, after a generation the algorithm will evolve to $k^1 = (k_{1best}^1, k_{2best}^1)$. After several generations, the algorithm can converge to the pure NE, i.e. the intersection of the best response curves as shown in Figure 3.6. No generator can find a better variable to alternate the variable selected previously so the generators have no incentive to change their variables.

When the transmission constraint is tighter, the limit cycle phenomenon presented by Weber [35] is observed in Figure 3.7. For example, at generation 2, the best solution of CGA evolves to k^2 and at generation 3, it is attracted to k^3 because Generator 2 has the highest profit in k^3 . It then evolves to k^4 at generation 4 and k^5 at generation 5. Finally, it evolves and returns to k^2 at generation 6.

CGA is a global search algorithm and can avoid converging to a local optimal point as illustrated by the simulation results.



Figure 3.6: Convergence process for $\overline{F} = 180$ MW



Figure 3.7: Convergence process for $\overline{F} = 80$ MW

B Effect of Generation Capacity Constraints:

When the generator hits its generation capacity constraints, it cannot use its bidding variable to change its output to get more profit. The bidding variables' threshold can be determined by:

$$k_{Gi\,Lim} = \lambda \,/(a_i P_{G\,iLim} + b_i) \qquad i = 1, \dots, N \tag{3.14}$$

where k_{GiLim} is equal to k_{Gimin} or k_{Gimax} when P_{GiLim} is equal to $\underline{P_{Gi}}$ or $\overline{P_{Gi}}$ respectively. On the one hand, if the output of a generator reaches its lower capacity limit ($\underline{P_{Gi}}$), the generator can select its bidding variable not less than the threshold (k_{Gimin}) when other generators do not change their strategies. On the other hand, if its output reaches the upper limit ($\overline{P_{Gi}}$), it can select the bidding variable not larger than the threshold (k_{Gimax}) when other generators do not change their strategies. An equilibrium continuum may exist in such situation.

Figure 3.8 shows there exists a NE continuum because the best response curve of Generator 1 has much intersection with response curve of Generator 2 when the market constraints are $\overline{F} = 180$ and $\overline{P_{G2}} = 50$ MW. Generator 2 hits its upper capacity constraint and can select the bidding variable arbitrarily and no larger than $k_{G2max} = 1.70$ (Point A in Figure 3.8) if Generator 1 does not change its strategy $k_{G1} = 1.60$. But if Generator 2 selects k_{G2} between 1.56 and 1.70 as its bidding strategy, Generator 1 will not satisfy its benefit and changes its strategy. For example, if $k_{G2} = 1.70$, k_{G1} will become 1.30 (Point B in Figure 3.8). The equilibrium continuum is $k_{G1} = 1.60$ and $k_{G2} \in [0, 1.56]$ as shown in Figure 3.8. The NE continuum upper bound $k_{G2max}^{(c)}$ of k_{G2} is smaller than k_{Gimax} . Figure 3.9 shows when $\overline{F} = 180$ and $\overline{P_{G2}} = 50$ MW, Generator 2

can select the bidding variable arbitrarily not larger than the $k_{G2max}^{(c)}$. However, the output of Generator 1 and Generator 2 are unchanged after 20 generations as shown in Figure 3.10.

As shown in Figure 3.8, one pure NE (Point D) in the continuum is converged stochastically in the proposed algorithm in this case. The NE continuum also exists in LSFE under *a*-parametrization and *b*-parametrization with generation capacity constraint $\overline{P_{G2}}$ =50MW.



Figure 3.8: Best response curves for $\overline{F} = 180$ MW and $\overline{P_{G2}} = 50$ MW.



Figure 3.9: Evolution process of bidding variables for \overline{F} =180MW and $\overline{P_{G2}}$ =50MW.



Figure 3.10: Evolution process of output for \overline{F} =180MW and $\overline{P_{G_2}}$ =50MW.

C Simulation results:

In this section, the impact of transmission and generation capacity constraints on the behavior of generators are also investigated. LSFE with different supply function parameters and Cournot model are tested. The upper and lower limits of bidding variable are assumed to be a_i *20, b_i *20, 20 and 0, 0, 0 in LSFE under *a*-parametrization, *b*-parametrization and ($a \propto b$) -parametrization respectively. The upper and lower limits of α_i and β_i are assumed to be a_i *10⁶ and $-a_i$ *10⁶, b_i *10⁶ and $-b_i$ *10⁶ in LSFE under (a, b) -parametrization test cases respectively.

The following four cases are studied:

- Case A: $\overline{F} = 180$ MW
- Case B: $\overline{F} = 80$ MW
- Case C: $\overline{F} = 180$ MW and $\overline{P_{G2}} = 50$ MW
- Case D: $\overline{F} = 80$ MW and $\overline{P_{G2}} = 50$ MW

The simulation results of different cases are shown in Table 3.3. It shows that the profits of the GenCos are more than the profits in the competitive case because each GenCo can exercise its market power through the strategic bidding to yield higher price and profit in Case A. The results of different LSFE types depend on the method of parametrization greatly. Different LSFE parametrization will result in different solutions. How to select the proper coefficients for the LSFE is very important in market analysis.

The market may have no pure NE in LSFE under *a*-, *b*-, and $(a \propto b)$ -parametrization if the transmission line is tight (\overline{F} =50MW) as shown in Case B. However, when the generators also have tighter generation capacity constraints, the system can have a pure NE due to the change of power flow pattern as shown in Case D.

Basing on the comparisons between Cases A and C, the market price increases due to the generation capacity constraints. For example, in Case C, if the power output of Generator 2 reaches its upper limit, Generator 1 can withhold some generating abilities to lead the shortage of power supply and lifts up the market price. The market has a NE continuum due to the capacity constraints.

In the LSFE under (a, b) -parametrization simulation cases, i.e. Cases A, B, C and D, the slope and intercept of supply functions are not unique, but the market-clearing price and the generators' outputs are unique and equal to the results of the Cournot model in Cases A, B, C and D respectively.

In these cases, the slope of the supply function is a very large positive value and the intercept is a very large negative value when the capacity constraints are not hit. Since the slopes become large, the generators become less competitive and their bidding behaviors are similar to that of Cournot quantity decision makers. A focal equilibrium [95] is found under (a,b)-parametrization because it is mutually beneficial to all the strategic generators in Case A. The simulation results demonstrate that it can also converge to the focal equilibrium in tighter capacity and transmission constrained markets in Cases B, C and D.

When Generator 2 hits its capacity constraint, its output cannot be changed. Its behavior is similar to that of a Cournot quantity decision price maker. As there are only two generators in the system, the residual demand is provided by the Generator 1 only.

58

The bidding behaviors of Generator 1 and the quantity decision market price maker are also alike. The outcomes of LSFE in Cases C and D are close to that of the Cournot model. Tight generation capacity constraint may make LSFE market equilibrium approach to Cournot market equilibrium.

	Case A						
Results		LSFE	LSFE	LSFE	LSFE		
	Perfect	under	under	under	under	Cour.	
	comp.	<i>a</i> -para.	<i>b</i> -para.	$(a \propto b)$ -para.	(a,b)-para		
α_1 or β_1 or k_{G1}		0.0512	11.633	1.1502			
$\alpha_2 \operatorname{or}_{\beta_2} or_{\beta_2} or k_{G2}$		0.0512	11.632	1.1502			
$P_{G11}(MW)$	111.11	94.68	102.02	101.08	76.92	76.92	
P_{G22} (MW)	111.11	94.68	102.07	101.08	76.92	76.92	
π_{l} (\$/h)	123.46	369.63	270.64	284.71	532.54	532.54	
$\pi_2($/h)$	123.46	369.63	270.72	284.71	532.54	532.54	
<i>p</i> ₁ (\$/MWh)	12.22	14.85	13.67	13.83	17.69	17.69	
<i>p</i> ₂ (\$/MWh)	12.22	14.85	13.67	13.83	17.69	17.69	
			Ca	ase B			
Paculte	Deutent	LSFE	LSFE	LSFE	LSFE		
Results	Perfect	under	under	under	under	Cour.	
	comp.	<i>a</i> -para.	<i>b</i> -para.	$(a \propto b)$ -para.	(a,b) -para.		
α_1 or β_1 or k_{GI}							
$\alpha_2 \operatorname{or} \beta_2 \operatorname{or} k_{G2}$							
$P_{G11}(MW)$	80.00				76.92	76.92	
$P_{G22}(MW)$	136.00				76.92	76.92	
π_{l} (\$/h)	64.00				532.54	532.54	
$\pi_2(h)$	184.96				532.55	532.54	
<i>p</i> ₁ (\$/MWh)	11.60				17.69	17.69	
<i>p</i> ₂ (\$/MWh)	12.72				17.69	17.69	
_		Case C					
Paculte	Perfect	LSFE	LSFE	LSFE	LSFE		
Results		under	under	under	under	Cour.	
	comp.	<i>a</i> -para.	<i>b</i> -para.	$(a \propto b)$ -para.	(a,b)-para		
α_1 or β_1 or k_{GI}		0.1000	17.111	1.6038			
$\alpha_2 \operatorname{or} \beta_2 \operatorname{or} k_{G2}$		*	*	*			
$P_{G11}(MW)$	160.00	88.89	88.89	88.89	88.89	88.89	
P_{G22} (MW)	50.00	50.00	50.00	50.00	50.00	50.00	
$\pi_{l}(h)$	256.00	711.11	711.11	711.11	711.11	711.11	

Table 3.3: Case study results of 2-bus test system
$\pi_2($/h)$	135.00	419.44	419.44	419.44	419.44	419.44
<i>p</i> ₁ (\$/MWh)	13.20	18.89	18.89	18.89	18.89	18.89
<i>p</i> ₂ (\$/MWh)	13.20	18.89	18.89	18.89	18.89	18.89
			Ca	se D		
Results	Perfect	LSFE	LSFE	LSFE	LSFE	
Results		under	under	under	under	Cour.
	comp.	<i>a</i> -para.	<i>b</i> -para.	$(a \propto b)$ -para.	(a,b) -para.	
α_1 or β_1 or k_{GI}		0.120	18.000	1.6897		
$\alpha_2 \operatorname{or}_{\beta_2} \operatorname{or}_{\beta_2} k_{G2}$		*	*	*		
$P_{G11}(MW)$	80.00	80.00	80.00	80.00	80.00	80.00
$P_{G22}(MW)$	50.00	50.00	50.00	50.00	50.00	50.00
π_{l} (\$/h)	64.00	704.00	704.00	704.00	704.01	704.00
$\pi_2(\text{h})$	455.00	455.00	455.00	455.00	455.00	455.00
<i>p</i> ₁ (\$/MWh)	11.60	19.60	19.60	19.60	19.60	19.60
<i>p</i> ₂ (\$/MWh)	19.60	19.60	19.60	19.60	19.60	19.60

Notes:

1. '*' means that the bidding variable is not unique and there exist a continuum.

2. The bold characters mean that the capacity constraint of the generator is hit.

3.3.2 IEEE 30-Bus Test System

A modified IEEE-30 bus system from [96] is applied in this example. The IEEE 30-bus test system has 30 buses, 41 lines, 6 generators and 21 loads. The transmission data can be obtained in [96]. The 6 generators belong to two GenCos. Generators 1, 2, 3 on buses 1, 2 and 5 respectively belong to GenCo 1 and Generators 4, 5, 6 on buses 8, 11 and 13 respectively belong to GenCo 2. Tight transfer limits are assumed in the transmission line between buses 1 and 3, 2 and 4, 2 and 6, 5 and 7 (branch 2, 3, 6 and 8 in [96] respectively). The upper and lower limits of all bidding variables are same as the cases in the 2-bus test system respectively. The coefficients of generators' cost function are given in Table 3.4. The demand curve of each demand is $p_i = 45 - d_i P_{Di}$ where d_i is

chosen such that $p_i = 35 \,(\text{MWh})$ for the assumed value of P_{Di} in [96]. The demand constraints are $\underline{P_{Di}} = 0$ MW and $\overline{P_{Di}} = 45/d_i$ MW.

Generator	Bus	$a_{\rm I}({\rm (MW)}^{2}{\rm h})$	$b_{\rm c}$ (\$/MWh)	$c_{.}(\$/h)$	P_{Ci} (MW)	$\overline{P_{\alpha}}$ (MW)	
No.	No.					G_{l}	
1	1	0.075	20	0	50	200	
2	2	0.35	17.5	0	20	80	
3	5	1.25	10	0	15	50	
4	8	0.1668	32.5	0	10	35	
5	11	0.5	30	0	10	30	
6	13	0.5	30	0	12	40	

Table 3.4: Coefficients of generators' cost function

The following three cases are studied:

Case A: the transmission constraints of all branches are equal to the original data in [96]

Case B: the transmission constraints of branches 2, 3, 6 and 8 are 50% of the original data

Case C: the transmission constraints of branches 2, 3, 6 and 8 are 20% of the original data

The simulation results are shown in Table 3.5. The results of LSFE under $(a \propto b)$ -parametrization are similar to that of *a*-parametrization and *b*-parametrization in Cases A and C. Besides, no pure NE is found in Case B due to transmission constraints. Therefore, these results are not shown. In Cases A and C, there may exist a NE continuum because generators hit their generation capacity constraints. The algorithm can converge to one pure NE stochastically.

As shown in Table 3.5, the market price in LSFE competition is higher than that in perfect competition in Cases A and C respectively. In the LSFE under

(a,b)-parametrization in Cases A and C, the slope and intercept of supply functions are not unique, but the market-clearing price and the generators' outputs are unique and equal (except the minor difference) to the results of the Cournot model in Cases A and C respectively. It means that a focal equilibrium is converged. The minor numerical difference of solutions in Table 3.5 is mainly due to the limits of 10⁶ order for the limits of supply function parameters α_i and β_i . The numerical difference will be reduced when larger limitation values of supply function parameters are used.

The transmission constraint can make the difference of nodal price in Case C. Generators 4, 5 and 6 can generate more power so GenCo 2 gets more profit. The profit of GenCo 1 decreases because transmission constraints make its generators (Generators 1, 2 and 3) impossible to generate more power. Due to different locations in the network, some generators can increase their outputs for getting higher profits due to the congestion; and the other generators may lose some market power and get lower profits. Therefore, it is clear that the generators may not increase their profits for the cases of congestion.

Because transmission constraints and generation capacity constraints make generators become less competitive, the solutions of different parametrization LSFE are similar to Cournot model in Cases A and C. When the transmission constraints become tighter, the market power becomes more serious in Case C. It can be observed that the obtained results and phenomena of this test system are consistent with that of the 2-bus system.

		Cas	se C	
Results	Darfaat aamn	LSFE under	LSFE under	Cour
	Perfect comp.	$(a \propto b)$ -para.	(<i>a</i> , <i>b</i>)-para.	Cour.
k_{G1}		1.1848		
k_{G2}		1.1915		
k_{G3}		1.2123		
k_{G4}		1.0268		
k_{G5}		*		
k_{G6}		*		
$P_{G11}(MW)$	193.3829	144.7677	136.5854	136.3446
$P_{G22}(MW)$	48.7485	37.6687	36.2161	36.3362
$P_{G35}(MW)$	19.6448	16.1261	15.9037	16.1773
$P_{G48}(MW)$	12.2750	18.5977	17.8328	17.8193
$P_{G511}(MW)$	10.0000	10.0000	10.9451	10.8954
$P_{G613}(MW)$	12.0000	12.0000	12.0000	12.0000
$\pi_l(10^{^3})/h)$	2.0595	2.3469	2.3518	2.3523
$\pi_2(10^{^3})/h)$	0.0516	0.1299	0.1443	0.1441
<i>p</i> ₁ (\$/MWh)	34.5037	36.5601	36.9016	36.8984
<i>p</i> ₂ (\$/MWh)	34.562	36.5601	36.9016	36.8984
<i>p</i> ₅ (\$/MWh)	34.5560	36.5601	36.9016	36.8984
p_8 (\$/MWh)	34.5499	36.5601	36.9016	36.8984
<i>p</i> ₁₁ (\$/MWh)	34.5493	36.5601	36.9016	36.8984
<i>p</i> ₁₃ (\$/MWh)	34.5477	36.5601	36.9016	36.8984
		Cas	se C	
Results	Parfact comp	LSFE under	LSFE under	Cour
	r encer comp.	$(a \propto b)$ -para.	(<i>a</i> , <i>b</i>) -para.	Cour.
k_{GI}		1.6148		
k_{G2}		*		
k_{G3}		1.2482		
k_{G4}		1.0922		
k_{G5}		1.1027		
k_{G6}		*		
$P_{GII}(MW)$	98.4109	58.1776	56.6242	57.0328
$P_{G22}(MW)$	24.8167	20.0000	20.0000	20.000
$P_{G35}(MW)$	22.0147	17.4258	17.6490	17.7344
$P_{G48}(MW)$	35.0000	25.4502	23.7169	23.7633
$P_{G511}(MW)$	19.4192	12.7261	13.4194	13.3774
$P_{G613}(MW)$	16.6885	12.0000	14.7066	14.5566
$\pi_I(10^{^3})$	0.7739	1.6896	1.6586	1.6857
$\pi_2(10^{^3})/h)$	0.3326	0.3122	0.3120	0.3125
<i>p</i> ₁ (\$/MWh)	27.3808	39.3419	39.9931	39.4656
<i>p</i> ₂ (\$/MWh)	26.1858	39.2013	39.9729	39.3655

Table 3.5: Case study results of IEEE 30-bus test system

<i>p</i> ₅ (\$/MWh)	37.5184	39.6706	39.9238	39.6996
<i>p</i> ₈ (\$/MWh)	40.2413	40.1386	39.9443	40.0328
<i>p</i> ₁₁ (\$/MWh)	39.7096	40.0976	39.9568	40.0036
<i>p</i> ₁₃ (\$/MWh)	38.3443	39.9923	39.9931	39.9287

Notes:

- 1. '*' means that the bidding variable is not unique and there may exist a continuum.
- 2. The bold characters mean that the capacity constraint of the generator is hit.

3.4 Conclusion

This chapter has successfully applied CGA to determine the market equilibrium in electricity market with generation capacity and transmission constraints under the LSFE model and the Cournot model.

The simulation results show that the LSFE under (a,b)-parametrization converges to a focal equilibrium that is equal to Cournot equilibrium under the capacity and transition transmission constraints in all cases. The simulation results have also shown that the system may not have a pure equilibrium due to the transmission congestion. But an equilibrium continuum may exist when the generation capacity of generators becomes tighter in LSFE under *a*-, *b*- and $(a \propto b)$ -parametrization. The solutions of market power analysis are different and depending on the parameters selected in different types of LSFE. It is also demonstrated that when these constraints are hit, the market price will increase and the abuse of market power can become more serious. Tighter constraints lead the solution of LSFE under *a*-, *b*- and $(a \propto b)$ -parametrization close to the solution of Cournot model.

The results also show that Cournot prices are possible outcomes in the markets where bids can be changed in one pricing period (i.e. an hour). These observations should be contrasted with the multiple pricing period models. The requirement to bid consistently across a time horizon may limit the exercise of market power in above multiple pricing-period models. It is important to study electricity market in multiple pricing period market.

CHAPTER 4 MULTI-PERIOD MARKET SIMULATION AND ANALYSIS

4.1 Introduction

In the study of Chapter 3, a supply function is specified for each hour-long pricing period. Recently, some studies have been performed on determination of the market equilibrium with LSFE in a single pricing period [34-36]. This assumption is reasonable in several BBP markets, such as England and Wales market after 2001 and California power market. However, this assumption is not a feature of all BBP markets. In several BBP markets, such as the England and Wales market until 2001 and PJM market (PJM interconnection is a regional transmission organization that coordinates the movement of wholesale electricity in all or parts of Delaware, Illinois, Indiana, Kentucky, Maryland, Michigan, New Jersey, North Carolina, Ohio, Pennsylvania, Tennessee, Virginia, West Virginia and the District of Columbia), a single supply function is applied across multiple pricing period. Green and Newbery [25] model each player by specifying a single supply function bid that is applied to all pricing period over an extended length of time. This assumption will be investigated in this chapter.

Baldick and Hogan show that SFE prices will be below Cournot prices if it is constant through a time horizon [91]. The market outcome of a supply function across all pricing period should be compared with that of the supply function in a single pricing period, and whether it can give important information to market regulators and policy makers should be studied. It is also important to find a bidding rule for the regulator and whether it is useful to relieve the market power.

Compared to actual markets, most of these studies have not considered the inter-temporal constraints of the real market. But the regulators and market participants need a new methodology to analyze market power and bidding strategies in multiple pricing period market. It is desirable for market simulation to reproduce as close as possible the actual market functioning to make the simulator results comparable to actual market settings. Many researches have also paid attention to bidding behavior of GenCos and Unit Commitment (UC) problem in a market environment [97]. Multi-period electricity auction market simulation method that explicitly takes into account transmission congestion as well as inter-temporal operating constraints such as start-up costs, ramp rates etc is presented in [98, 99].

To determine the market equilibrium is generally modeled as an EPEC, such as transmission line constraints and capacity constraints, are considered. It will become more complex to model the effect of the actual GenCos on market-clearing prices in multi-period market setting considering the inter-temporal constraints. Co-evolutionary computation approach has successfully used to economical simulation [33, 55-58] and it will be employed as an alternative tool to simulate the complex electricity market game.

The purpose of this chapter is to provide a co-evolutionary computation approach to simulate a day-ahead BBP electricity market in which GenCos are with capacity and inter-temporal constraints. The outcomes of a market in which GenCos use the supply functions being constant or non-constant across multiple pricing period are studied and compared.

4.2 Market Model Formulation

Suppose there are *I* GenCos, each has a thermal unit (i.e. generator), in the test system. The GenCo is assumed to submit a bid for its unit to the ISO, in the form of LSF, including the intercept and slope of the LSF. Each unit is assumed to have a strictly concave quadratic cost function:

$$C_i(P_{G_i}(t)) = \frac{1}{2}a_i P_{G_i}^{\ 2}(t) + b_i P_{G_i}(t) + c_i \quad i = 1, \dots, I$$
(4.1)

where *I* is number of units; $P_{Gi}(t)$ is power generated by unit *i* at time *t*; $C_i(P_{Gi}(t))$ is fuel cost of unit *i* for generating power at time *t*; a_i, b_i and c_i are unit *i* cost coefficients.

The marginal cost function is affine for each unit as follows:

$$\frac{dC_i}{dP_{G_i}}(P_{G_i}(t)) = a_i P_{G_i}(t) + b_i \quad i = 1, \dots, I$$
(4.2)

It is assumed that the system demand has an inverse linear demand function and the transmission loss is negligible. Then the system demand at pricing period t will be equal to the total output of all the units in the market as shown below:

$$P_{S}(t) = \overline{P_{S}(t)} - rp(t) = \sum_{i=1}^{I} P_{Gi}(t) \quad i = 1, ..., I$$
(4.3)

where $P_s(t)$ is system demand at time t; r and $\overline{P_s(t)}$ are coefficients of demand function at time t, r is the slope of demand function and $\overline{P_s(t)}$ is the forecasted demand at time t.

If $r \neq 0$, the system demand benefit function can be represented by:

$$B(P_{s}(t)) = -\frac{1}{2r}P_{s}^{2}(t) + \frac{1}{r}\overline{P_{s}(t)}P_{s}(t)$$
(4.4)

 $B(P_s(t))$ is the benefit of system demand at t.

The GenCos need to provide their bids to the power market with an inverse LSF form as follows.

$$p(t) = \alpha_i(t)P_{G_i}(t) + \beta_i(t) \quad i = 1, ..., I$$
(4.5)

where α_i and β_i are LSF coefficients.

In a day-ahead power market, the GenCos submit the bidding sets to the ISO. The transactions are determined by the UC and Economic Dispatch (ED). The UC optimization problem can be formulated as the following mixed integer programming problem. The ISO optimization problem is formulated as to minimize total fuel cost (i.e. (4.6)) or to maximize the summation of social welfare (i.e. (4.7)) when the slope of the demand function is equal to zero or not respectively.

$$Min \quad F = \sum_{t=1}^{T} \left\{ \sum_{i=1}^{I} U_i(t) \left(\frac{1}{2} \alpha_i(t) P_{G_i}^2(t) + \beta_i(t) P_{G_i}(t) \right) + S_i(x_i(t-1), U_i(t)) \right\}$$
(4.6)

$$Max \quad \Gamma_{ISO} = \sum_{t=1}^{T} \left(-\frac{1}{2r} P_s^2(t) + \frac{1}{r} \overline{P_s(t)} P_s(t) \right) - \sum_{t=1}^{T} \left\{ \sum_{i=1}^{I} U_i(t) \left(\frac{1}{2} \alpha_i(t) P_{Gi}^2(t) + \beta_i(t) P_{Gi}(t) \right) + S_i(x_i(t-1), U_i(t)) \right\}$$
(4.7)

s.t.
$$\sum_{i=1}^{l} P_{Gi}(t) U_i(t) = P_s(t) \quad \text{energy balance}$$
(4.8)

$$\sum_{i=1}^{l} \overline{P_{Gi}} U_i(t) \ge R(t) + P_S(t) \quad \text{spinning reserve}$$
(4.9)

$$\underline{P_{Gi}} \le P_{Gi}(t) \le \overline{P_{Gi}} \quad \text{capacity constraint} \tag{4.10}$$

$$U_{i}(t) = 1 \quad if \quad 1 \le x_{i}(t) \le \tau_{i} \quad \text{min up time}$$

$$U_{i}(t) = 0 \quad if \quad -\underline{\tau_{i}} \le x_{i}(t) \le -1 \quad \text{min down time}$$
(4.11)

where T is time horizon; $x_i(t)$ is state of unit *i* at time *t*, denoting number of hours that the unit has been on (positive) or off (negative); $U_i(t)$ is decision variable of unit *i* at time *t*, 1 for up, 0 for down; $S_i(x_i(t-1), U_i(t))$ is start-up cost of unit *i* at time *t*; R(t) is system spinning reserve.

The GenCos' optimization problem can be formulated as (4.12) and the decision variable are $\alpha_i(t)$ and $\beta_i(t)$.

$$Max \quad \pi_i = \sum_{t=1}^{T} (p(t)P_{Gi}(t) - C_i(P_{Gi}(t)) - S_i(x_i(t-1), U_i(t)))$$
(4.12)

So (4.6)-(4.12) formulate a two-level optimization problem. The first level optimization problem is UC and ED optimization problems determined by ISO; and the second level optimization problem is each market GenCo to maximize its profit. This equilibrium model can be categorized as an EPEC.

4.2.1 Single pricing period market model

First the minimum up and down time constraints and start-up cost are not considered and all units must be turned on in each pricing period. The ISO optimization problem formulated in (4.7) can be reformed as (4.13) in each pricing period.

$$Max \quad \Gamma_{ISO}(t) = -\frac{1}{2r}P_s^2(t) + \frac{1}{r}\overline{P_s(t)}P_s(t) - \sum_{i=1}^{I} \left(\frac{1}{2}\alpha_i(t)P_{Gi}^2(t) + \beta_i(t)P_{Gi}(t)\right) (4.13)$$

The optimization problem of the GenCo *i* in each pricing period *t* is formulated as:

$$\pi_{i}(t) = p(t)P_{Gi}(t) - C_{i}(P_{Gi}(t))$$

= $p(t)\left(P_{S}(t) - \sum_{j \neq i} P_{Gj}(t)\right) - C_{i}\left(P_{S}(t) - \sum_{j \neq i} P_{Gj}(t)\right)$ (4.14)

Differentiating the profit with respect to price at time *t* yields:

$$\frac{d\pi_{i}(t)}{dp(t)} = \left(P_{S}(t) - \sum_{j \neq i} P_{Gj}(t)\right) + \left(p(t) - C_{i}'\left(P_{S}(t) - \sum_{j \neq i} P_{Gj}(t)\right)\right) \times \left(\frac{dP_{S}(t)}{dp} - \sum_{j \neq i} \frac{dP_{Gj}(t)}{dp}\right)$$
(4.15)

Setting this derivative to zero produces:

$$P_{Gi}(t) = \left(p(t) - C_i'\left(P_S(t) - \sum_{j \neq i} P_{Gj}(t)\right)\right) \times \left(-\frac{dP_S(t)}{dp} + \sum_{j \neq i} \frac{dP_{Gj}(t)}{dp}\right)$$
(4.16)

 $P_{Gi}(t)$ will not be equal to zero in a single pricing period. If $P_{Gi}(t)$ is equal to zero, GenCo *i* will not satisfy its profit and will change its strategy. Then other GenCos will also change their strategies.

4.2.2 Multiple pricing period market model

The minimum up and down time constraints and start-up cost are not considered and all units must be turned on in each pricing period too. Assume the bid supply function must be consistent across all pricing period.

$$\begin{cases} \alpha_i(t) = \alpha_i(1) & t = 2, \dots, T \\ \beta_i(t) = \beta_i(1) \end{cases}$$
(4.17)

This assumption matches with several markets, such as England and Wales until 2001 and markets in the Eastern United States such as PJM.

If $P_{Gi}(t)$ is not equal to zero, (4.12) can be differentiated with respect to p(t).

$$\frac{d\pi_i}{dp(t)} = \frac{d[\sum_{t=1}^{l} \pi_i(t)]}{dp(t)} = \frac{d\pi_i(t)}{dp(t)}$$
(4.18)

Setting this derivative to zero, (4.16) is also obtained. Because the bid supply functions are consistent across all times, these equations must be satisfied at every realized value of price *p*. Therefore, (4.19) can be obtained under this assumption [30].

$$\begin{cases} \beta_{i}(t) = b_{i} & i = 1, \dots, I, \quad t = 1, \dots, T\\ \alpha_{i}(t) = \frac{1}{(r + \sum_{i=1, j \neq i}^{l} \frac{1}{\alpha_{j}})} + a_{i} & (4.19) \end{cases}$$

In a low demand period, if $P_{Gi}(t)$ is equal to zero, p(t) is nonsensical to the profit of supplier *i*. Then profit function (4.12) cannot be differentiated with respect to p(t). It means (4.16) (i.e. equation (4) in [26]) may not exist and (4.19) may not be obtained. It is difficult to analyze the condition for the existence of market equilibrium when capacity constraints are considered. In this chapter, a co-evolutionary computation approach is employed to find the NE under these situations.

4.2.3 Market equilibrium with inter-temporal constraints

When the inter-temporal constraints are considered in the market clearing model, the GenCos' optimization problems defined in (4.12) are very complex. It is hard to acquire the NE by analytical methods, so the CGA is used as an alternative tool to market simulation.

The UC optimization problem is solved by Dynamic Programming (DP) methods. DP is characterized by the forward and back path operations. Commitment of units progresses one hour at a time, and combinations of schedulable units are stored for each hour. Finally, the most economical schedule is obtained by backtracking from the combination with the least total cost at the final hour through the optimal path to the combination at the initial hour. Thus if there are I schedulable units in an hour, then there are 2^{I} combinations to evaluate. Obviously, it is not practical to evaluate all of the combinations.

The Dynamic Programming-Sequential Combination (DP-SC) method [100] generates a subset of the combinations by turning each unit on in priority list sequence. Thus if there are I schedulable units, only I+1 combinations will be evaluated. Proper ordering of the units in the priority list is expected to yield more economic schedules. Generators with lower hear rate will be at higher priority. If generators have the same heat rate (same HR_i value), the one with higher capacity ($\overline{P_{Gi}}$) will be of higher priority. The calculation for HR is given:

$$HR_{i} = \frac{C_{i}(\overline{P_{Gi}})}{\overline{P_{Gi}}} \quad i = 1, \dots, I$$

$$(4.20)$$

Since many combinations are neglected, the optimal path may not be found. This technique seems to be well suited when the system load is changing rapidly. DP-SC method is presented in [100].

4.3 Co-evolutionary Approach to Analyzing Market Equilibrium

The basic cooperative coevolution model is given in this Chapter 2. The co-evolutionary computation model can be regarded as a special form of the Agent-based Computational Economics (ACE) model [101]. ACE is a computational study of economies modeled as dynamic systems of interacting agents. The ability of ACE to capture the independent decision-making behavior and interactions of individual agents provides a very good platform for the modeling and simulation of electricity markets. Each GenCo in the market under investigation is represented by an agent in the model.

The co-evolutionary algorithm comprises multiple species (i.e. agents). Each agent represents a GenCo in the market. The agents interact with one another within a shared domain model. The market clearing problem is modeled in the domain model. In the algorithm, agents can calculate their best strategies based on information about what other agents have done in the last generation [79]. Individuals in the agents with higher fitness are at a productive advantage compared to individuals with low fitness, hence the latter decrease in frequency in the population over time. When there exists a pure NE, the algorithm can find it efficiently [33, 56]. Each species has no incentive to change it strategy to get a higher fitness when the algorithm gets to the pure NE. When there is no pure NE, the algorithm could not find a stable strategies set. It means the algorithm could not find a strategy set that all the players satisfy their profits. The algorithm can find a Pareto solution under this condition [73].

The proposed approach is implemented using a real-coded GA. A decision variable x_i is represented by a real number within its lower limit *Lb* and its upper limit *Ub*, i.e. $x_i \in [Lb, Ub]$. The crossover and mutation operators are described as follows [102]:

Crossover:

A blend crossover operator has been employed in the study. This operator generates the offspring randomly from the interval $[x_i - \tau(y_i - x_i), y_i + \tau(y_i - x_i)]$, where x_i and y_i are the decision values of the parent solutions and $x_i < y_i$. τ is a random number between 0 and 1. To ensure the balance between exploitation and exploration of the search space $\tau = 0.5$ is selected.

Mutation:

A dynamical non-uniform mutation operator to reduce the disadvantage of random mutation in the real-coded GA is used and defined as follows:

$$x'_{i} = \begin{cases} x_{i} + \Delta(t, UB - x_{i}) & \text{if random } \tau \text{ is } 1\\ x_{i} - \Delta(t, LB - x_{i}) & \text{if random } \tau \text{ is } 0 \end{cases}$$
(4.21)

where x_i and x_i are the selected and resultant values respectively for the mutation; t is generation number of CGA. The function $\Delta(t, y)$ returns a value in the range [0, y]such that $\Delta(t, y)$ approaches to zero as t increases. This property causes this operator to search the space uniformly in the initial stages (i.e. when t is small), and locally at later stages. This strategy increases the probability of generating a new number close to its successor than a random choice. The following function is used:

$$\Delta(t, y) = y(1 - \sigma^{(1 - \frac{t}{T})^{o}})$$
(4.22)

where σ is a uniform random number from [0, 1], *T* is the maximal generation number, and δ is a system parameter determining the degree of dependency on the iteration number. In order to prevent the algorithm from stalling or premature convergence, it needs to explore the search space when the generations become large. So $\delta = 1$ is used to replace the generally used $\delta = 5$.

Fitness scaling technical is also employed to improve performance of the algorithm.

Fitness scaling:

At the start of GA runs, it is common to have a few individuals of extraordinary fitness, while the rest of the population has mediocre fitness. If left to normal selection, these few extraordinary strings will take over a significant proportion of the finite population. This usually leads to premature convergence. There is another problem that occurs quite late in the runs. The population average fitness may be close to the best fitness of population. And the survival of the fittest which is so important for improvement across generations just becomes a random walk among the mediocre. In both cases, fitness scaling can be helpful. A fitness scaling technique based on simulated anneal technique is given:

$$f_{j}' = e^{f_{j}/T_{em}}$$
 (4.23a)

$$T_{em} = T_{em}^0(0.99^{t-1}) \tag{4.23b}$$

where T_{em}^{0} is initial temperature; T_{em} is the temperature; f_{j} is fitness of individual j; f_{j} is scaled fitness of individual j;

The flow chart of CGA is shown in Figure 4.1.



Figure 4.1: Flow chart of the CGA

4.4 Case Study

In this chapter, the proposed method is applied to revised IEEE 6 generators test system [103]. The 6 generators belong to 6 GenCos respectively. The CGA parameters are given in Table 4.1 and each generation contains 2 nested generations of evolution of each species.

Parameters	Description
Variable code type	Real code
Initial population	Randomly initialized population
Population size	40
Max. generation	300
Mutation	Non-uniform mutation(mutation probability= 0.05, $\delta = 1$)
Crossover	Blend crossover (crossover probability= 0.9)
Selection	Roulette selection with fitness scaling (T_{em}^0 =500) and 5% population
	are elitism

Table 4.1: CGA parameters

Load demand data is shown in Table 4.2.

Hour	$\overline{P_s(t)}$ (MW)	Hour	$\overline{P_s(t)}(\mathrm{MW})$	Hour	$\overline{P_s(t)}$ (MW)
1	166	9	192	17	246
2	196	10	161	18	241
3	229	11	147	19	236
4	267	12	160	20	225
5	283.4	13	170	21	204
6	272	14	185	22	182
7	246	15	208	23	161
8	213	16	232	24	131

Table 4.2: Load demands for 24 hours

The coefficient of 6 generators study cases are given in Table 4.3.

GenCos No.	1	2	3	4	5	6
$\underline{P_{Gi}}(MW)$	50	20	15	10	10	12
$\overline{P_{Gi}}(MW)$	200	80	50	35	30	40
$a_i (10^{-2} \text{/(WM)}^2 \text{h})$	0.75	3.5	12.5	1.668	5.0	5.0
$b_i(\text{WM})h)$	2	1.75	1	3.25	3	3
$c_i($)$	0	0	0	0	0	0
Min up time(Hr)	1	2	1	1	2	1
Min down time(Hr)	1	2	1	2	1	1
Init unit status	1	3	2	3	2	2
Start cost (\$)	123.0	130.5	81.5	188.5	126.0	76.5

Table 4.3: Coefficients of GenCos

The study cased and simulation results are showed in Tables 4.4 and 4.5.

Case A₁: single pricing period LSFE model

Case B₁: single pricing period Cournot model

Case C₁: single pricing period LSFE with capacity constraints.

Case D₁: single pricing period Cournot model with capacity constraints.

Case E₁: multi pricing period LSFE model

Case F₁: multi pricing period LSFE with true cost coefficient β_i and capacity constraints.

Case G₁: multi pricing period LSFE with true cost coefficient β_i and inter-temporal constraints.

Cases A₂, B₂, C₂, D₂, E₂, F₂ and G₂ using same data but the loads increase 80MW in each pricing period comparing to Cases A₁, B₁, C₁, D₁, E₁, F₁ and G₁ respectively.

The load elastic is r=5 MW/\$ in all pricing period. The study cases with reserve requirement ($R_s(t)=0.1P_s(t)$)are studied in Cases G₁ and G₂.

In the single period LSFE simulation cases, i.e. Cases A₁, C₁, A₂ and C₂, the slope and intercept of supply functions are not unique, but the market price and the GenCos' outputs are unique and equal to the results of the Cournot model in Cases B₁, D₁, B₂ and D₂ respectively. The minor numerical difference of solutions in Table III is mainly due to the limits of 10⁴ order for the limits of supply function parameters α_i and β_i . The numerical difference will be reduced when larger limitation values of supply function is a very large positive value and the intercept is a very large negative value when the capacity constraints are not hit. The simulation results demonstrate that it can converge to the focal equilibrium which equal to Cournot NE [33, 34].

By comparing Case A_1 with C_1 , when there has upper capacity constraint, the market price is rising and GenCos can have more market power in the peak demand pricing period (i.e. hour 5). Because some GenCos hit the upper capacity constraints, it leads to high price as shown in Figure 4.2. Due to the same reason, the market price in the peak demand periods (i.e. hours 3-8 and 15-21) with capacity constraints is higher than that without capacity constraints by comparing Cases A_2 with C_2 as shown in Figure 4.4.

The market price is higher in the case that the GenCos without lower capacity constraints than that in the case with lower capacity constraints by comparing Cases A_1 and C_1 in hours 1-4 and 6-24. In these hours, the system demand is low and output of some GenCos is also very low because they have no lower capacity constraints in Case A_1 . It means that the GenCos can withhold their output to uplift the market price, so the market price is higher in Case A_1 than in Case C_1 in these hours respectively. Due to the

same reason, the market price will be higher in hours 1, 3-8 and 15-21 in Case A_2 than that in Case C_2 .

In Case G₂, the bidding variable of β_i is equal to b_i . This result is equal to that the GenCos are assumed to bid β_i honestly. The seemingly simple result is important since no other strategic hypothesis is imposed but the general genetic operators on the GenCos' strategic learning, and only the multiple pricing period bidding requirement settles the supply functions down on a unique equilibrium. This is a key notion of SFE theories [30].

In another multiple period Case G₁, the bidding variable of β_i is not equal to b_i because the output of some GenCos is equal to zero in several valley load periods, such as in hours 10-12 and 24. Although β_i is not equal to b_i exactly, it is close to b_i as shown in Table 4.4.

The NE existence and uniqueness condition is very hard to be obtained in a multiple pricing period market with capacity constraint. There may be multiple equilibriums. Simulation results show there exists multiple equilibriums. The algorithm can converge to one pure NE stochastically in each run.

Some of these equilibriums are Pareto efficient [84]. In these equilibriums, the GenCos hold their output and lift the market prices which are close to the Cournot market prices emerging in a single pricing period market. While the others aren't Pareto efficient, the outcome of the market is close to the outcome of the market in which the GenCos are assumed to bid β_i honestly. Baldick and Hogan use the stability theory to analyze the multiple pricing period market and point out only the equilibria that are

82

Case	A_1	B_1	C_1	D_1	E_1	F_1	G_1
α_1/a_1					1.9091	2.6209	
α_{2}/a_{2}					1.1252	1.2726	
α_{3}/a_{3}					1.0253	1.0658	
$\alpha_{_4}/a_{_4}$					1.4171	1.4833	
α_{5}/a_{5}					1.1154	1.1463	
$\alpha_{_6}/a_{_6}$					1.1131	1.1452	
β_1/b_1					1.0936	1.0000	
β_2/b_2					1.0515	1.0000	
β_3/b_3					1.0498	1.0000	
eta $_{_4}$ /b $_{_4}$					0.9999	1.0000	
β_5/b_5					1.0007	1.0000	
eta $_6$ /b $_6$					1.0010	1.0000	
<i>p</i> (1)(\$/MWh)	7.5507	7.5571	6.6138	6.6277	3.3162	3.2813	
<i>p</i> (2)(\$/MWh)	8.5367	8.5585	7.8163	7.8288	3.4773	3.5847	
<i>p</i> (3)(\$/MWh)	9.6732	9.6875	9.1384	9.1542	3.6545	3.8006	
<i>P</i> (4)(\$/MWh)	10.950	10.935	10.696	10.708	3.8585	4.0363	
<i>p</i> (5)(\$/MWh)	11.531	11.580	11.670	11.674	3.9465	4.1466	
<i>P</i> (6)(\$/MWh)	11.118	11.118	10.967	10.979	3.8853	4.0675	
<i>P</i> (7)(\$/MWh)	10.244	10.139	9.7966	9.8330	3.7457	3.9061	
<i>p</i> (8)(\$/MWh)	9.1357	9.1288	8.4992	8.5103	3.5685	3.7015	
<i>p</i> (9)(\$/MWh)	8.4303	8.4201	7.6554	7.6687	3.4558	3.5550	
<i>p</i> (10)(\$/MWh)	7.3889	7.3883	6.4199	6.4256	3.2893	3.2231	
<i>p</i> (11)(\$/MWh)	6.9267	6.9207	5.6534	5.6554	3.2037	3.0593	
<i>p</i> (12)(\$/MWh)	7.4007	7.4128	6.3774	6.3830	3.284	3.2115	
<i>P</i> (13)(\$/MWh)	7.6913	7.6912	6.7751	6.7937	3.3377	3.3282	
<i>p</i> (14)(\$/MWh)	8.1952	8.1395	7.3751	7.3883	3.4182	3.5001	
<i>P</i> (15)(\$/MWh)	8.9677	8.9849	8.2987	8.3099	3.5417	3.6684	
<i>p</i> (16)(\$/MWh)	9.7740	9.7291	9.2574	9.2792	3.6706	3.8193	
<i>p</i> (17)(\$/MWh)	10.244	10.245	9.7950	9.8968	3.7457	3.9061	
<i>p</i> (18)(\$/MWh)	10.076	10.076	9.6176	9.6327	3.7189	3.8752	
<i>P</i> (19)(\$/MWh)	9.9084	9.9094	9.4018	9.4329	3.692	3.8441	
<i>p</i> (20)(\$/MWh)	9.5389	9.5391	8.9782	8.9913	3.633	3.7758	
<i>P</i> (21)(\$/MWh)	8.8334	8.8354	8.1154	8.1548	3.5202	3.6406	
<i>P</i> (22)(\$/MWh)	8.0933	8.0946	7.254	7.2649	3.4021	3.4683	
<i>p</i> (23)(\$/MWh)	7.5188	7.5292	6.4133	6.4245	3.2893	3.2231	
<i>p</i> (24)(\$/MWh)	6.3808	6.3815	2.8000	2.8000	3.0925	2.6443	
$\pi_1(10^{^3}\$)$	5.6503	5.6657	7.1242	7.1658	2.7184	2.6428	
$\pi_2(10^{^3}\$)$	5.0735	5.0992	4.2858	4.3170	1.1394	1.2153	
$\pi_3(10^{^3}\$)$	3.8615	3.8725	3.2612	3.2863	0.6187	0.6556	
$\pi_4(10^{^3}\$)$	3.6122	3.6122	2.9203	2.9220	0.0842	0.1320	
$\pi_5(10^{^3}\$)$	3.1544	3.1564	2.5559	2.5659	0.0788	0.1008	
$\pi_6(10^{3})$	3.1632	3.1874	2.5918	2.6104	0.0789	0.0950	

Table 4.4: Simulation results of low demand cases

close to competitive market equilibria are stable [91]. Then the supply functions with true coefficient β_i are used in this chapter to study the multiple pricing period market.

The price curve of the multiple periods market is smoother compared to the price curve of a single period market because each GenCo considers not only benefit of the high demand periods but also benefit of the low demand periods. It will causes price of the high demand periods to be low and price of the low demand periods to be high as shown in Figures 4.2 and 4.4. And the market price in multiple pricing period is also lower than the price of single pricing period market respectively.

In Case G_1 , there may be no pure NE, the algorithm could not find a strategy set that it is satisfied by all GenCos. The algorithm can get a Pareto optimal solution in each run. 20 Pareto solutions will be obtained by running the algorithm 20 times as shown in Figure 4.3. Although there is no pure NE in Case G_1 , the market price is fluctuating in a small range except pricing period 1. In Case G_2 , the system demand is very high and the algorithm can find a pure NE. The market price in Cases G_1 and G_2 are close to in Cases F_1 and F_2 and is lower that that in Cases D_1 and D_2 respectively because the GenCos are not dispatched if their bids are too high in Cases G_1 and G_2 . Therefore, the market power abuse can be mitigated.

It is found that the requirement of consistent biding across multiple pricing period can generate lower market price and alleviate the exercise of market power. Since some electricity markets have been set up with an obligation for consistent bids while others have not, it is important to confirm that the impacts of market rule specification on the outcomes.

Case	A_2	B2	C_2	D_2	E ₂	F2	G ₂
α_1/a_1					2.1879	2.6998	2.8434
α_2/a_2					1.1902	1.2499	1.2915
α_{3}/a_{3}					1.0448	1.0657	1.0710
$\alpha_{_4}/a_{_4}$					1.4474	1.5085	1.4944
α_{5}/a_{5}					1.1328	1.1809	1.1858
α_{6}/a_{6}					1.1244	1.1861	1.2011
β_1/b_1					0.9941	1.0000	1.0000
β_2/b_2					0.9998	1.0000	1.0000
β_3/b_3					1.0092	1.0000	1.0000
eta $_{_4}$ /b $_{_4}$					1.0005	1.0000	1.0000
β_5/b_5					0.9985	1.0000	1.0000
β_{6}/b_{6}					1.0010	1.0000	1.0000
<i>p</i> (1) (\$/MWh)	10.2107	10.2323	9.8062	9.8331	3.7639	3.9275	3.9699
<i>p</i> (2) (\$/MWh)	11.3089	11.3206	11.2166	11.218	3.9358	4.1170	4.1787
<i>p</i> (3) (\$/MWh)	12.3413	12.3537	13.0321	13.057	4.1250	4.3908	4.4650
<i>P</i> (4) (\$/MWh)	13.4322	13.6383	15.2445	15.273	4.3428	4.7113	4.7992
<i>p</i> (5) (\$/MWh)	14.1594	14.1887	16.1947	16.215	4.4368	4.8627	4.9657
<i>P</i> (6) (\$/MWh)	13.8032	13.8141	15.5198	15.564	4.3714	4.7535	4.8494
<i>P</i> (7) (\$/MWh)	12.7309	12.7619	14.0169	14.048	4.2224	4.5342	4.6128
<i>p</i> (8) (\$/MWh)	11.8123	11.8236	12.2565	12.265	4.0333	4.2558	4.3260
<i>p</i> (9) (\$/MWh)	11.1089	11.1136	10.9713	10.979	3.9129	4.0917	4.1436
<i>p</i> (10) (\$/MWh)	10.0376	10.0762	9.6156	9.6326	3.7352	3.8959	3.9376
<i>p</i> (11) (\$/MWh)	9.5567	9.6061	9.0539	9.0711	3.6550	3.8075	3.8473
<i>p</i> (12) (\$/MWh)	10.0498	10.0472	9.3001	9.5926	3.7295	3.8896	3.9312
<i>P</i> (13) (\$/MWh)	10.3470	10.379	9.7934	9.9934	3.7868	3.9528	3.9957
<i>p</i> (14) (\$/MWh)	10.8476	10.8584	10.6004	10.613	3.8728	4.0475	4.0922
<i>P</i> (15) (\$/MWh)	11.6503	11.6558	11.9501	11.959	4.0046	4.2137	4.2824
<i>p</i> (16) (\$/MWh)	12.4309	12.4618	13.1966	13.232	4.1422	4.4161	4.4911
<i>p</i> (17) (\$/MWh)	12.9109	12.9322	14.0213	14.048	4.2224	4.5342	4.6128
<i>p</i> (18) (\$/MWh)	12.7244	12.7639	13.7166	13.756	4.1938	4.4920	4.5691
<i>P</i> (19) (\$/MWh)	12.5634	12.5962	13.4402	13.465	4.1651	4.4499	4.5261
<i>p</i> (20) (\$/MWh)	12.2308	12.227	12.9904	12.707	4.1020	4.3571	4.4301
<i>P</i> (21) (\$/MWh)	11.5104	11.5278	11.7746	11.714	3.9817	4.1799	4.2475
<i>P</i> (22) (\$/MWh)	10.7243	10.7448	10.4540	10.474	3.8556	4.0286	4.0730
<i>p</i> (23) (\$/MWh)	10.1002	10.0711	9.3527	9.6581	3.7352	3.8959	3.9376
<i>p</i> (24) (\$/MWh)	9.0768	9.0688	8.4142	8.4302	3.5633	3.7058	3.7442
$\pi_1(10^{^3}\$)$	10.645	10.742	12.706	12.772	4.5812	4.8959	4.9103
$\pi_2(10^{^3}\$)$	9.4082	9.427	10.347	10.397	1.7842	2.1386	2.1497
$\pi_3(10^{^3}\$)$	6.7980	6.822	7.3950	7.427	0.8669	1.0069	1.0469
$\pi_4(10^{^3}\$)$	7.5781	7.609	6.8590	6.800	0.3997	0.5677	0.6223
$\pi_5(10^{^3}\$)$	6.5070	6.524	5.7240	5.762	0.2495	0.3769	0.3435
$\pi_6(10^{3})$	6.5394	6.545	6.8300	6.854	0.2495	0.3769	0.4157

Table 4.5: Simulation results of high demand cases



Figure 4.2: Market price of lower demand cases



Figure 4.3: Market price of lower demand case with inter-temporal constraints



Figure 4.4: Market price of higher demand cases

4.5 Conclusion

A co-evolutionary computation approach has been applied to analyze market equilibrium in multiple pricing period electricity market in this chapter. The capacity constraints and inter-temporal constraints are considered in the market model. It is found that when ISO compels the GenCos with a supply function being constant across multiple pricing period, the market price will be lower than that of Cournot price emerging in a single pricing period market and the market power can also be alleviated. Besides, it is important to further study the outcomes of different markets in which the GenCos are compelled to bid consistent or not for multiple pricing period. It is also important to observe outcomes and to set up an obligation for consistent bids or not.

CHAPTER 5 ANALYZING TWO-SETTLEMENT ELECTRICITY MARKET EQUILIBRIUM

5.1 Introduction

Due to California electricity crisis, more and more researchers are convinced that the forward contract plays an important role for market power mitigation [3] or risk management [104] in electricity markets. Generally, there are at least three main reasons for the GenCos to enter a forward market: 1) being compelled by regulators as part of a transitional vesting process, which is the situation in many countries, such as England and Wales, Australia and New Zealand; 2) recognizing the importance and necessity of risk management with contractual arrangements [104]; and 3) being encouraged economically [105, 106].

The electricity spot markets modeled by SFE and Cournot models have been extended to include contract markets in [23, 27, 45, 52, 107-109]. The English electricity market is modeled by SFE model with a contract market, and the entry condition of the contract market is discussed in [107]. Reference [27] has showed that competition in contract market could lead the generators to sell contracts and increase their outputs and also hedge the spot market price in England and Wales. Reference [108] proposes an asymmetric LSFE model to develop firms' optimal bidding strategies given their forward contracts. Market power mitigation effects of forward contracts

89

have also been evaluated. Reference [109] examines the issue of whether generators would voluntarily enter contract markets through an economic incentive by a two-stage game model proposed in [105]. The contract market is modeled with the general conjectural variation method. The factors which could affect this strategic contracting behavior are investigated. The dynamics in two-settlement electricity markets are studied by an agent-based model. Numerical simulations illustrate that the access to the forward market leads to more competitive behaviors of the suppliers in the spot market, and thus lower spot energy prices [45]. References [23, 52], determination of the equilibrium in two-settlement electricity markets is formulated as an EPEC using Cournot model, in which each firm solves a MPEC. It is shown that spot market prices will decreases when the supplies enter forward contracts.

The main objective of this chapter is to investigate the issue of whether generators would voluntarily enter the contract market solely through the economic incentive and to examine the factors that could affect the strategic contracting behavior. Therefore, market equilibrium models are required to simulate the competition of generators in spot and forward markets. However, determination of the market equilibrium will be difficult when the forward market and generation capacity constraints are considered. Co-evolutionary computation is successfully used to determine the equilibrium of noncooperative game and market simulation. It can handle the nonlinear market models that are difficult to be handled by conventional methods. Thus, this chapter will employ the co-evolutionary computation to study bidding behavior of the GenCos in a two-settlement market. The impact of forward contracts on market outcomes is then

90

analyzed. The entry condition of GenCos in forward markets and the factors which could affect the bidding behavior are also studied.

5.2 Market Model Formulation

The real electricity market is a multi-settlement market where forward transactions, day-ahead transactions, and real-time balancing transactions are settled sequentially at different prices. It can be simplified as a two-settlement market including a forward market and a spot market. This chapter uses the two-stage game model in [105] to formulate the two-settlement market consisting of a forward market and a spot market. In the first stage, GenCos enter the forward market, forming rational expectations regarding the forward contracts of their rivals and the spot equilibrium outcomes. In the second stage, the GenCos compete in the spot market taking into account all GenCos' forward contracts. The Cournot or LSFE models are used to represent competitive bidding in the spot market, while the forward market is represented by the Cournot model.

5.2.1 Market assumptions

Suppose there are *I* GenCos (suppliers) in the electricity market. And these GenCos are risk neutral. Each GenCo has a generator and is characterized by the following quadratic cost function:

$$C_{i} = \frac{1}{2}a_{i}P_{Gi}^{2} + b_{i}P_{Gi} \quad i = 1, \dots, I$$

$$a_{i} > 0$$
(5.1)

where P_{Gi} is the quantity generated by Generator *i*; a_i and b_i are the coefficients of the Generator *i*'s cost function. The marginal cost function of Generator *i* is affine as follows:

$$MC_{i} = \frac{dC_{i}}{dP_{G_{i}}} = a_{i}P_{G_{i}} + b_{i} \quad i = 1, \dots, I$$
(5.2)

When there is negligible transmission loss, the aggregate demand P_s , which is assumed to be an inverse linear demand function, will be equal to the total output of all GenCos as shown below:

$$P_{S} = \overline{P_{S}} - rp = \sum_{i=1}^{I} P_{G_{i}} \quad i = 1, ..., I$$
 (5.3)

where *p* is the spot market price; *r* and $\overline{P_s}$ are coefficients of the demand function. And *r* is the slope of the demand function.

The GenCos compete with each other in the forward market by choosing the quantity of their contracts P_{Gi}^c ($P_{Gi}^c \ge 0$), which they are willing to sell at the forward market price p_i^c . It is assumed that the forward contracts are observable for all the GenCos in spot market. Also the GenCos are assumed to use their bidding strategies, such as quantities of generator in Cournot model and coefficients of supply functions in LSFE model, to compete in the spot market.

5.2.2 Equilibrium model for strategic bidding in the spot market

The GenCos bid in the spot market by using the information observed in the forward market. LSFE model and the less competitive market model, Cournot model, are employed to model the spot market respectively in this section. Given the decisions of GenCos in the forward market, the optimization problem faced by each GenCo is how to maximize its expected total profit (π_{Gi}), expressed below.

$$\pi_{Gi} = p(P_{Gi} - P_{Gi}^c) + p_i^c P_{Gi}^c - C_i$$
(5.4)

1) Cournot model:

In Cournot model, each GenCo will maximize its own profit in (5.4) by changing its decision variable P_{Gi} . The derivative of GenCo *i*'s profit with respect to its decision variable can be written as:

$$\frac{\partial \pi_{Gi}}{\partial P_{Gi}} = (P_{Gi} - P_{Gi}^c) \frac{\partial p}{\partial P_{Gi}} + p - MC_i$$
(5.5)

By setting the derivative of GenCo *i*'s profit in (5.5) as zero and eliminating p using (5.3), the optimal value of P_{Gi} can be obtained by:

$$P_{Gi} = \frac{(\overline{P_S} - \sum_{i=1}^{l} P_{Gi} + P_{Gi}^c - b_i r)}{(1 + a_i r)} = \frac{(rp + P_{Gi}^c - b_i r)}{(1 + a_i r)}$$
(5.6)

Then, by eliminating P_{Gi} in (5.6) using (5.3), the spot market price can be determined by:

$$p = \frac{1}{r + \sum_{i=1}^{I} \frac{r}{1 + a_i r}} \left[\overline{P_s} - \sum_{i=1}^{I} \frac{1}{1 + a_i r} P_{G_i}^c + r \sum_{i=1}^{I} \frac{b_i}{1 + a_i r} \right]$$
(5.7)

2) LSFE model:

The GenCos are assumed to compete in the spot market by submitting their bids in the form of LSF. The supply function is a non-decreasing function of price and takes the form as follows:

$$p = \alpha_i P_{Gi} + \beta_i \quad i = 1, ..., I$$

$$\alpha_i > 0 \tag{5.8}$$

where α_i and β_i are the coefficients of LSF.

LSFs with *a*-, *b*-, $(a \propto b)$ - and (a,b)-parametrization introduced in chapter 3 are studied in this chapter. The equation (5.9) is used to study (a,b)-parametrization in this chapter.

$$p_n(P_{Gin}) = k_{Gi}(a_i P_{Gin} + b_i) \quad i = 1, \dots, I$$
(5.9)

In LSFE model, each GenCo will maximize its own profit expressed in (5.4) by changing its decision variables α_i , β_i or k_{Gi} according to its parametrization type. By differentiating (5.4) with respect to p and using (5.3) and (5.8), the derivative of GenCo *i*'s profit with respect to p can be written as:

$$\frac{\partial \pi_{Gi}}{\partial p} = (P_{Gi} - P_{Gi}^{c}) + p \frac{\partial P_{Gi}}{\partial p} - MC_{i} \frac{\partial P_{Gi}}{\partial p}$$

$$= (P_{Gi} - P_{Gi}^{c}) + (p - MC_{i}) \frac{\partial (\overline{P_{S}} - rp - \sum_{j=1, j \neq i}^{l} \frac{p - \beta_{i}}{\alpha_{i}})}{\partial p}$$
(5.10)

By setting the derivative of GenCo *i*'s profit in (5.10) as zero, the optimal value of P_{Gi} can be obtained by:

$$P_{Gi} = P_{Gi}^{c} + (p - MC_{i})(r + \sum_{j=1, j \neq i}^{l} \frac{1}{\alpha_{j}})$$
(5.11)

5.2.3 Equilibrium model for strategic contracting in the forward market

The equilibrium for the forward market can be determined by maximizing GenCos' profits with nesting the equilibrium conditions from the spot market in the calculation. In the forward market, all GenCos are supposed to be able to offer forward contracts. It is also assumed that there are enough risk neutral arbitrageurs in the markets and they will eliminate any profitable arbitrage opportunity arising from the difference between

the forward prices and the expected spot prices. Then p_i^c is an unbiased estimator of p [105].

Each GenCo chooses its forward market output so as to maximize its profit, which can be formulated as follows:

$$Max \quad \pi_{Gi} = p(P_{Gi} - P_{Gi}^{c}) + p_{i}^{c} P_{Gi}^{c} - C_{i}$$
(5.12)
s.t. $p_{i}^{c} = p \quad i = 1, ..., I$

Since the spot price p is an implicit function of the forward market output P_{Gi}^c , the GenCo *i*'s marginal benefit for forward contract is equal to the derivative of its profit with respect to P_{Gi}^c as follows:

$$\frac{\partial \pi_{G_i}}{\partial P_{G_i}^c} = \frac{\partial p}{\partial P_{G_i}^c} P_{G_i} + (p - MC_i) \frac{\partial P_{G_i}}{\partial P_{G_i}^c}$$
(5.13)

A Modeling spot market by Cournot model:

Substituting (6) into (13) yields:

$$\frac{\partial \pi_{Gi}}{\partial P_{Gi}^c} = \frac{\partial p}{\partial P_{Gi}^c} P_{Gi} + (p - MC_i) \left(\frac{r}{1 + a_i r} \frac{\partial p}{\partial P_{Gi}^c} + \frac{1}{1 + a_i r}\right)$$
(5.14)

Then, setting the marginal benefit in (5.14) to zero and using (5.7), the GenCo i's output in the spot market can be obtained as follows:

$$P_{Gi} = (p - MC_i)(r + \sum_{j=1, j \neq i}^{I} \frac{r}{1 + a_j r})$$
(5.15)

By eliminating p in (5.15) using (5.6), the GenCo *i*'s output in forward market can be written as:
$$P_{Gi}^{c} = \frac{\sum_{j=1, j\neq i}^{I} \frac{1}{1+a_{j}r}}{1+\sum_{j=1, j\neq i}^{I} \frac{1}{1+a_{j}r}} P_{Gi}$$
(5.16)

The fraction of power sold at the forward market in (5.16) and spot market in (5.15) for each GenCo can be determined by:

$$\frac{P_{Gi}^{c}}{P_{Gi}} = \frac{\sum_{j=1, j\neq i}^{I} \frac{1}{1+a_{j}r}}{1+\sum_{j=1, j\neq i}^{I} \frac{1}{1+a_{j}r}} = 1 - \frac{1}{1+\sum_{j=1, j\neq i}^{I} \frac{1}{1+a_{j}r}}$$
(5.17)

When r is increasing, the demand becomes more elastic. The GenCos will have less incentive to lift up the market price in spot market. They will bid aggressively (low price for large quantities) in spot market and have less incentive to enter forward market. Then the fraction in (5.17) will become smaller.

It can also be observed that the fraction is affected by the slope of GenCos' marginal cost functions. If a_i is larger than a_i (for any *i* and $i \in 1,...,I$), $\sum_{j=1,j\neq i}^{I} \frac{1}{1+a_jr}$ is larger

than
$$\sum_{j=1,j\neq i'}^{l} \frac{1}{1+a_j r}$$
. According to (5.17), $\frac{P_{Gi}^c}{P_{Gi}} > \frac{P_{Gi}^c}{P_{Gi}}$ can be obtained. It means that a

GenCo with a large slope of marginal cost function will have more incentive to enter the forward market.

B Modeling spot market by LSFE model:

 $\frac{\partial P_{G_i}^c}{\partial P_{G_i}}$ is obtained by substituting (5.8) into (5.11) and differentiating it with respect to

 P_{Gi} as follows:

$$\frac{\partial P_{Gi}^c}{\partial P_{Gi}} = \frac{\partial [P_{Gi} - [(\alpha_i P_{Gi} + \beta_i) - (a_i P_{Gi} + b_i)]A_i]}{\partial P_{Gi}} = 1 + (a_i - \alpha_i)A_i$$
(5.18)

where $A_i = (r + \sum_{j=1, j \neq i}^{I} \frac{1}{\alpha_j}) > 0$.

Substituting (5.3) and (5.8) into (5.11) and differentiating it with respect to p yields:

$$\frac{\partial P_{Gi}^{c}}{\partial p} = \partial \{\overline{(P_{S} - rp - \sum_{j=1, j \neq i}^{I} \frac{p - \beta_{i}}{\alpha_{i}}}) - [(\alpha_{i} - a_{i})\overline{(P_{S} - rp - \sum_{j=1, j \neq i}^{I} \frac{p - \beta_{i}}{\alpha_{i}}}) + (\beta_{i} - b_{i})]A_{i}\}/\partial p$$

$$= -[1 + (a_{i} - \alpha_{i})A_{i}]A_{i}$$
(5.19)

Substituting (5.18) and (5.19) into (5.13) produces:

$$\frac{\partial \pi_{Gi}}{\partial P_{Gi}^{c}} = \frac{-1}{[1 + ((a_{i} - \alpha_{i})A_{i}]A_{i}}P_{Gi} + \frac{(p - MC_{i})}{1 + (a_{i} - \alpha_{i})A_{i}} \\
= \frac{-P_{Gi} + (p - MC_{i})A_{i}}{[1 + (a_{i} - \alpha_{i})A_{i}]A_{i}}$$
(5.20)

By substituting (5.11) to (5.20), the marginal benefit for the forward contract can be obtained by

$$\frac{\partial \pi_{G_i}}{\partial P_{G_i}^c} = \frac{-P_{G_i}^c}{\left[1 + (a_i - \alpha_i)A_i\right]A_i}$$
(5.21)

1) For a, b or $(a \propto b)$ -parametrization of LSF:

By substituting (5.2) and (5.8) to (5.11), the generation output in spot market becomes:

$$P_{Gi} - P_{Gi}^{c} = (p - MC_{i})A_{i} = [(\alpha_{i} - a_{i})P_{Gi} + (\beta_{i} - b_{i})]A_{i}$$
(5.22)

(5.23a) and (5.23b) are obtained by using (5.11), (5.21) and (5.22) when P_{Gi}^c is not equal to or equal to P_{Gi} respectively.

$$1 + (a_{i} - \alpha_{i})A_{i} = 1 + (a_{i} - \alpha_{i})\frac{P_{Gi} - P_{Gi}^{c}}{(p - MC_{i})}$$

$$= 1 + (a_{i} - \alpha_{i})\frac{P_{Gi}}{(p - MC_{i})} + (\alpha_{i} - a_{i})\frac{P_{Gi}^{c}}{(p - MC_{i})}$$

$$= \frac{\beta_{i} - b_{i}}{(p - MC_{i})} + \frac{(\alpha_{i} - a_{i})P_{Gi}^{c}}{(p - MC_{i})}$$

$$1 + (a_{i} - \alpha_{i})A_{i} = 1$$
(5.23b)

I) For $P_{Gi}^{c} = 0$:

 $\frac{\partial \pi_{Gi}}{\partial P_{Gi}^c} = 0 \text{ can be obtained according to (5.21).}$

II) For $0 < P_{Gi}^c < P_{Gi}$:

Using (5.21) and LSFE parametrization condition yields:

$$\begin{cases}
p > MC_i \\
\alpha_i \ge a_i \\
\beta_i \ge b_i \\
\alpha_i - a_i + \beta_i - b_i > 0
\end{cases}$$
(5.24)

 $\frac{\partial \pi_{Gi}}{\partial P_{Gi}^c}$ is less than zero according to (5.21), (5.23a) and (5.24).

III) For $P_{Gi}^c = P_{Gi}$:

 $\frac{\partial \pi_{Gi}}{\partial P_{Gi}^c}$ is less than zero according to (5.21) and (5.23b).

IV) For $P_{Gi} < P_{Gi}^c$:

Using (5.21) and LSFE parametrization condition yields:

$$\begin{cases} p < MC_i \\ \alpha_i \le a_i \\ \beta_i \le b_i \\ \alpha_i - a_i + \beta_i - b_i < 0 \end{cases}$$
(5.25)

 $\frac{\partial \pi_{Gi}}{\partial P_{Gi}^c}$ is less than zero according to (5.21), (5.23a) and (5.25).

Under all the above conditions, GenCos will get negative marginal profit when it sells the forward contract so they have no incentive to enter the forward market.

2) For (a,b)-parametrization of LSF:

The GenCo may select a very large positive α_i as the supply function coefficient.

Then, $[1 + (a_i - \alpha_i)A_i]$ is less than zero and $\frac{\partial \pi_{G_i}}{\partial P_{G_i}^c}$ in (5.21) becomes larger than zero

so GenCos have incentive to enter the forward market

It is found that when GenCos decide their bidding variables using a, b and $(a \propto b)$ -parametrization of LSF, the spot market is more competitive. GenCos bid aggressively in spot market and have no incentive to enter forward market because their marginal benefits for forward contracts in the forward market are not larger than zero. However, GenCos may have incentive to enter forward market if they decide their supply functions using (a, b)-parametrization of LSF.

Determination of the market equilibrium in the two-stage game is a difficult task especially for the cases of the spot market under LSFE type competition or GenCos with tight capacity constraints. Therefore, a co-evolutionary computation approach is applied to solve this problem.

5.3 Co-evolutionary Computation Approach to Analyzing Market Model

In this chapter, CGA implemented using real-coded GAs is employed to determine the market equilibrium. The crossover and mutation operators of real-coded GA are described in Chapter 4. GenCos in the forward market and in spot market are represented by species in the outer loop and inner loop respectively in the CGA. So totally 2**I* species are needed. To determine the equilibrium of the two-stage game, the backward induction method [84] is merged into the CGA. It proceeds by considering the decisions might be made in spot market by GenCos by fixing forward contracts. Using the information obtained in spot market, GenCos can optimize their strategies in the forward market. Thus to solve the equilibrium of the forward market, the equilibrium of the spot market (i.e. subgame perfect equilibrium [84]) should be determined firstly.

The procedures can be summarized as follows:

- *Step 1:* Set the basic parameters of CGA including crossover rate, mutation rate, maximum generation number and population size in the outer loop and inner loop respectively.
- Step 2: The individual of each species, which includes the bidding variable of the GenCo in outer and inner loops, is initialized. All the initial representatives of each species are set randomly.
- *Step 3:* The real-coded GA is applied to solve the optimization problem of each GenCo for the forward market in (5.12). For each individual of the species in the outer loop, a fitness value is assigned according to its profit in (5.12). To get the fitness value, the forward market price, which is equal to the spot market price, is needed so the equilibrium of spot market and the corresponding price are determined first.
 - Step 3.1: The representatives in the outer loop (i.e. forward contracts of GenCos) are fixed and sent to the inner loop.

- Step 3.2: The profit of each GenCo for the spot market in (5.4) is maximized by the real-coded GA with consideration of its generation capacity constraint and system energy balance in (5.3). The corresponding bidding strategy (i.e. the best individual) is then determined. In the optimization, each individual of a species in the inner loop (i.e. a strategy of a GenCo in spot market) is evaluated and assigned a fitness value according to the corresponding profit in (5.4). The market price and total output of the GenCo in (5.4) can be obtained by (5.3) for Cournot model or (5.3) and (5.8) for LSFE model based on representatives of other species in the inner loop (i.e. strategies of other GenCos in spot market). The best individual of each species is the one with the highest fitness value.
- *Step 3.3:* The best individual of each species is selected as the representative in the inner loop and sent to other GenCos.
- Step 3.4: Repeat Steps 3.2 and 3.3 using the updated representatives and keeping the best strategy of each GenCo in the population (i.e. elitism) until the maximum generation number in the inner loop is reached.
- *Step 4:* The best individual of each species is selected as the representative of the species in the outer loop and sent to other GenCos.
- *Step 5:* Repeat Steps 3 and 4 using the updated representatives and keeping the best strategy of each GenCo in the population (i.e. elitism) until the maximum generation number is reached in the outer loop.

5.4 Case Study

In this section, a three GenCo test system in [109] and a five GenCo test system in [30] are used to validate the models and theoretical analyses in Section 5.2 and the effectiveness of the CGA in determination of market equilibrium described in Section 5.3. The CGA parameters are given in Table 5.1.

Parameters	Description	
Variable code type	Real code	
Initial population	Randomly initialized population	
Population size of outer loop	40	
Max. generation of outer loop	200	
Population size of inner loop	40	
Max. Max. generation of inner loop	40	
Mutation	Non-uniform mutation	
Mutation	(mutation probability = 0.05, $\delta = 1$)	
Crossover	Blend crossover	
Clossover	(crossover probability $= 0.9$)	
Solation	Tournament selection size is 2 and 5% population	
Selection	are elitism	

Table 5.1: CGA parameters

5.4.1 Three GenCos study case

The cost parameters of three GenCos in [109] are listed in Table 5.2. The aggregate demand parameters in (5.3) are $\overline{P_s(t)}$ =45GWh and *r*=0.5GWh/(\$/MWh).

Table 5.2: Cost coefficients of the three GenCos
--

GenCo No.	1	2	3
Cost parameter a_i \$/(MWh.GWh)	1.0	1.5	2.0
Cost parameter <i>b_i</i> \$/MWh	12.0	10.0	8.0

The following simulation cases with and without forward contracts are carried out:

Case A₁: Perfect competition in the spot market without forward market arrangement.

- Case A₂₁: Cournot type competition in the spot market without forward market arrangement.
- Case A₂₂: LSFE with *a*-parametrization type competition in the spot market without forward market arrangement.
- Case A_{23} : LSFE with *b*-parametrization type competition in the spot market without forward market arrangement.
- Case A₂₄: LSFE with $(a \propto b)$ -parametrization type competition in the spot market without forward market arrangement.
- Case A₂₅: LSFE with (a,b)-parametrization type competition in the spot market without forward market arrangement.
- Case A₃₁: Cournot type competition in the spot market with forward market arrangement.
- Case A₃₂: LSFE with *a*-parametrization type competition in the spot market with forward market arrangement.
- Case A_{33} : LSFE with *b*-parametrization type competition in the spot market with forward market arrangement.
- Case A₃₄: LSFE with $(a \propto b)$ -parametrization type competition in the spot market with forward market arrangement.
- Case A₃₅: LSFE with (a,b)-parametrization type competition in the spot market with forward market arrangement.

Simulation results are shown in Table 5.3. In Case A_{31} , the GenCos have the incentive to enter the forward market when the GenCos use the Cournot type competition in the

spot market. The solution of this simple case can be also obtained by solving (5.3), (5.7) and (5.15) using the symbolic tool in Matlab software [110].

By comparing Case A_{31} with Case A_{21} , it can be observed that GenCos prefer not to enter the forward market because they can get more profit without participating in the forward market. However, if one GenCo enters to the forward market, it could benefit from the forward market by producing more output. Other GenCos are prompted to enter the forward market by economic incentive. Finally, as shown in the market equilibrium of Case A_{31} , all GenCos enter the forward market, but they get less profit compared with the case of without participating in the forward market (i.e. Case A_{21}). The prisoners' dilemma-type outcome [105] appears in this situation.

In Case A₃₅, GenCos use LSFE with (a,b)-parametrization in the spot market, the slope and intercept of supply function of each GenCo are not unique, but the market-clearing price and the suppliers' outputs are unique and equal to the results of the Cournot model in Case A₃₁. In this situation, the slope of the supply function is a very large positive value and the intercept is a very large negative value. The GenCos become less competitive and behave like Cournot quantity decision-makers. The simulation result has demonstrated that a focal equilibrium [34] is converged.

When comparing Cases A_{35} and A_{31} with Cases A_{25} and A_{21} respectively, it is clear that the market price is lower and generation output increases when GenCos enter the forward market. The market becomes more competitive, and the market power is mitigated.

Case	Gen	$\alpha_{i/a} or \beta_{i/b} or k_{Gi}$	P_{Gi}^{c}	р	P_{Gi}	$\pi_{{\scriptscriptstyle Gi}}$
	No.	$/ a_i / b_i$ of	(GWh)	(\$/MWh)	(GWh)	(10^3\$)
	1				13.38	89.45
A_1	2			25.38	10.25	78.80
	3				8.69	75.47
	1				9.11	207.62
A_{21}	2			39.34	8.38	193.24
	3				7.83	184.15
	1	1.7501			10.97	150.55
A_{22}	2	1.4604		31.21	9.68	135.00
	3	1.3272			8.74	126.44
	1	1.5596			11.19	137.79
A ₂₃	2	1.4977		29.91	9.95	123.85
	3	1.5135			8.90	115.77
	1	1.3286			11.04	144.46
A_{24}	2	1.2375		30.61	9.82	130.05
	3	1.1921			8.84	121.68
A ₂₅	1				9.11	207.62
	2			39.34	8.38	193.24
	3				7.83	184.15
	1		5.48		10.59	164.45
A ₃₁	2		5.07	32.82	9.42	148.41
	3		4.75		8.58	139.33
	1	1.7501	0		10.97	150.55
A ₃₂	2	1.4604	0	31.21	9.68	135.00
	3	1.3272	0		8.74	126.44
	1	1.5596	0		11.19	137.79
A ₃₃	2	1.4977	0	29.91	9.95	123.85
	3	1.5135	0		8.90	115.77
	1	1.3286	0		11.04	144.46
A ₃₄	2	1.2375	0	30.61	9.82	130.05
	3	1.1921	0		8.84	121.68
	1		5.48		10.59	164.45
A ₃₅	2		5.07	32.82	9.42	148.41
	3		4.75		8.58	139.33

Table 5.3: Simulation results for three GenCos example

The simulation results in Cases A_{32} , A_{33} and A_{34} are the same as that in Cases A_{22} , A_{23} and A_{24} because the GenCos have no incentive to enter the forward market and no forward contract is arranged. It is consistent with the theoretical analysis in Section 5.2.

However, the spot market is still more competitive than that with forward contract arrangement in Cases A_{31} and A_{35} . It can be illustrated that the type of competitions and the parametrizations of LSF affect the GenCos' decisions and market equilibrium significantly.

The effects of the slope of the demand function on the bidding behavior are also studied and the simulation results are shown in Table 5.4. Cases B_1 and B_2 are same as Case A_{31} except for r=0.25 and r=1 respectively. It can be observed in Table 5.4 that the fraction of power sold at the forward market of each GenCo becomes smaller when r increases. Since the system demand is more elastic, the market is more competitive and the spot price reduces. The GenCos prefer to pursuit more profit in the spot market and reduce their output at the forward market. Similar situation can be obtained for Case A_{35} .

Case	Gen. No.	r GWh/ (\$/MWh)	P_{Gi}^c (GWh)	р (\$/MWh)	P _{Gi} (GWh)	P_{Gi}^{c} / P_{Gi}	π_{Gi} (10 ^{^3} \$)
	1		7.17		12.31	0.5825	329.03
B_1	2	0.25	6.64	44.88	11.17	0.5944	296.11
	3		6.22		10.29	0.6045	273.70
	1		5.48		10.59	0.5175	164.45
A ₃₁	2	0.50	5.07	32.82	9.42	0.5382	148.41
	3		4.75		8.58	0.5536	139.33
	1		3.24		7.66	0.4230	63.24
B_2	2	1.00	3.13	24.08	6.89	0.4543	61.42
	3		3.02		6.37	0.4741	61.87

Table 5.4: Simulation results for three GenCos example with different slopes of the

demand function

To investigate the effects of cost parameters on the bidding behavior, the following two cases are performed:

Cases C₁ and C₂ are same as Case A₃₁ except for $a_1 = 2$ and $b_1 = 8$ respectively.

The simulation results are shown in Table 5.5. In Case C₁, $a_1=a_3$, the fraction of GenCo 1 is the same as that of GenCo 3. And in Case C₂, $b_1=b_3<b_2$ and $a_1<a_2<a_3$, the fraction of GenCo 1 is smaller than that of GenCo 2 which is smaller than that of GenCo 3. It is not affected by the intercept of GenCos' marginal cost functions by comparing Case C₂ and Case A₃₁. It is observed that the fraction of power sold at the forward market is only affected by the slope of GenCos' marginal cost functions from Case C₁ and C₂.

Case	Gen.	P_{Gi}^{c}	р	P_{Gi}	P_{Gi}^{c}	$\pi_{\scriptscriptstyle Gi}$
	No.	(GWh)	(\$/MWh)	(GWh)	$/ P_{Gi}$	(10^3\$)
	1	5.48		10.59	0.5174	164.45
A ₃₁	2	5.07	32.82	9.42	0.5385	148.41
	3	4.75		8.58	0.5532	139.33
	1	4.08		7.89	0.5172	122.44
C_1	2	5.08	35.41	10.16	0.5000	180.73
	3	4.78		9.24	0.5172	167.87
	1	6.23		12.04	0.5174	212.53
C ₂	2	4.82	31.67	8.94	0.5385	133.81
	3	4.53		8.18	0.5532	126.70

Table 5.5: Simulation results for three GenCos example with different cost coefficients

The effects of capacity constraints on the bidding behavior are studied and the simulation results are shown in Table 5.6. Cases D_1 , D_2 , D_3 , D_4 and D_5 are same as Cases A_{31} , A_{32} , A_{33} , A_{34} and A_{35} except for $\overline{P_{G1}} = 5$ GWh. It can be observed that the market price will become higher when the capacity constraint is hit by comparing the simulation results of Cases D_1 , D_2 , D_3 , D_4 and D_5 in Table 5.6 with that of Cases A_{31} , A_{32} , A_{33} , A_{34} and A_{35} in Table 5.3 respectively. When GenCo 1 hits its capacity constraint, it could not increase its output to change the market price (*p*) and to pursuit

more profit. Then its expected profit in forward market defined in (5.12) (i.e. $p\overline{P_{G1}} - C_1(\overline{P_{G1}})$) is not affected by its forward contract output. The total output of GenCo 1 is unique and same as its maximum output of generation, but its output in the forward market is not unique as shown in Cases D₁, D₂, D₃, D₄ and D₅.

Case	Gen.	$\alpha_i / or \beta_i / or k_{Gi}$	P_{Gi}^c	p	P_{Gi}	$\pi_{_{Gi}}$
	No.	$/ u_i / v_i$	(GWh)	(\$/MWh)	(GWh)	(10^3\$)
	1		*		5.00	125.92
D_1	2		3.49	39.68	10.48	228.67
	3		3.52		9.68	213.01
	1	*	*		5.00	122.59
D_2	2	1.8159	0	39.02	10.65	224.03
	3	1.5766	0		9.84	208.36
	1	*	*		5.00	114.70
D ₃	2	2.0976	0	37.44	10.98	210.83
	3	2.1040	0		10.30	197.18
	1	*	*		5.00	120.13
D_4	2	1.4762	0	38.52	10.73	219.75
	3	1.3754	0		10.01	205.31
	1		*		5.00	125.92
D_5	2		3.49	39.68	10.48	228.67
	3		3.52		9.68	213.01

Table 5.6: Simulation results for three GenCos example with capacity constraints

** means the solution is not unique.

5.4.2 Five GenCos study case

A more realistic five GenCos test system in [30] is used to further validate the previous analysis. The test system is based on the cost data for the five strategic firms in England and Wales subsequent to the 1999 divestiture. The cost parameters of five GenCos are listed in Table 5.7. The aggregate demand parameters in (5.3) are $\overline{P_s(t)}$ =35GWh and *r*=0.1GWh/(ϵ /MWh).

GenCo No.	1	2	3	4	5
Cost parameter $a_i $ $\pounds/(MWh.GWh)$	2.687	4.615	1.789	1.930	4.615
Cost parameter $b_i $ $fmult MWh$	12	12	8	8	12

Table 5.8: Simulation results for five GenCos example

Table 5.7: Cost coefficients of the five GenCos

 P_{Gi}^{c} P_{Gi} р Gen $\alpha_i / a_i \text{ or } \beta_i / b_i \text{ or } k_{Gi}$ π_{Gi} Case No. (\pounds/MWh) (GWh) (GWh) (10^3£) 5.54 1 41.17 ----2 3.22 23.97 ---- E_1 3 26.87 99.57 10.55 ----4 9.78 92.30 ___ ____ 5 3.22 23.97 ___ ___ 1 5.39 329.66 ----2 269.57 4.68 -----3 E 21 80.40 6.14 410.86 ___ ___ 4 6.07 403.80 -----5 4.68 269.56 --___ 1 1.3106 5.92 76.32 --2 1.1672 3.87 46.11 --3 E 22 1.5035 32.85 9.24 153.18 ___ 4 144.06 1.4597 8.82 --5 1.1671 3.87 46.11 ___ 1 1.3051 5.90 68.27 --2 40.71 1.1776 3.76 --3 E 23 1.8282 31.50 9.43 142.09 ___ 4 8.99 1.7677 133.31 --5 1.1776 3.76 40.71 1 1.1594 5.90 73.01 --2 43.89 1.0889 3.83 --3 E 24 1.3090 32.30 9.32 148.79 ___ 4 1.2836 8.89 139.78 ___ 5 1.0889 3.83 43.89 --1 5.39 329.68 -----2 4.68 269.54 ----3 E 25 80.40 6.14 410.85 ----4 403.79 6.07 ----5 ___ --4.68 269.54 E 31 1 4.50 42.80 5.98 136.09 --2 3.33 4.39 90.71 --3 6.08 8.11 223.34 --

109

	4		5.90		7.86	213.92
	5		3.33		4.39	90.71
	1	1.3106	0		5.92	76.31
	2	1.1670	0		3.87	46.11
E 32	3	1.5033	0	32.85	9.24	153.17
	4	1.4597	0		8.82	144.05
	5	1.1672	0		3.87	46.10
	1	1.3050	0		5.90	68.27
	2	1.1776	0		3.76	40.71
E 33	3	1.8282	0	31.50	9.43	142.09
	4	1.7677	0		8.99	133.31
	5	1.1776	0		3.76	40.71
	1	1.1595	0		5.90	73.01
	2	1.0889	0		3.83	43.89
E 34	3	1.3090	0	32.30	9.32	148.79
	4	1.2837	0		8.89	139.78
	5	1.0890	0		3.83	43.89
	1		4.50		5.98	136.09
	2		3.33		4.39	90.71
E 35	3		6.08	42.80	8.11	223.34
	4		5.90		7.86	213.92
	5		3.33		4.39	90.71

Different simulation cases without and with forward contracts are carried out and the results are shown in Table 5.8. In this study, Cases E_1 , E_2 , E_{31} , E_{32} , E_{33} , E_{34} , E_{41} , E_{42} , E_{43} , E_{44} and E_5 are same as Cases A₁, A₂, A₃₁, A₃₂, A₃₃, A₃₄, A₄₁, A₄₂, A₄₃, A₄₄ and A₅ in three GenCos example respectively.

It can be observed that the simulation results of this five GenCos system are consistent with that of the three GenCos system. The prisoners' dilemma-type outcome appears in Case E_{31} and a focal equilibrium is converged in Case E_{35} . In Cases E_{32} , E_{33} and E_{34} , the GenCos have no incentive to enter the forward market. When comparing Case E_{35} and E_{31} with Case E_{25} and E_{21} respectively, it is also found that the forward market with Cournot type competition or LSFE type competition with (*a*,*b*)-parametrization can make the market more competitive and mitigate the market power.

5.5 Conclusion

A two-stage game model has been presented to model the two-settlement electricity market and to investigate whether a GenCo would enter forward market due to economic inspiration and which factors could affect their strategic behaviors. A co-evolutionary approach has been introduced and successfully employed to determine the market equilibrium of the two-stage game model in two numerical examples. Simulation results show that whether GenCos would enter the forward market depends significantly on what type of competitions in the spot market and whether generation capacity constraints are hit. When the forward market is under Cournot type competition or LSFE type competition with (a,b)-parametrization, GenCos may enter the forward market due to the economic incentive and also mitigate the market power. They may also enter forward market when their capacity constraints are hit. It is also found that their levels of participation in forward market depend significantly on the slope of the demand function and the slope of their marginal cost function.

CHAPTER 6 CONCUSION AND FUTURE WORK

6.1 Conclusion

Electricity restructuring has taken place in many countries to create competitive electricity markets. In these electricity markets, participants submit bids to a market pool that determines the market prices and transactions. This analysis of market equilibrium is of fundamental importance because it provides regulators with relevant information to identify and mitigate the exercise of market power. It also provides GenCos with the appropriate information to maximize their respective profits, within the regulatory framework, by altering market clearing prices to their own respective benefits.

The co-evolutionary computation is developed and employed to analyze the market equilibrium. The thesis establishes the mathematical models for single pricing period market, multi pricing period market and two-settlement market; investigates the convergence process of co-evolutionary computation; and uses co-evolutionary computation and game theories to analyze the equilibrium of electricity markets.

Firstly, the parallel CGA has been successfully applied to determine the market equilibrium in electricity market with generation capacity and transmission constraints under the LSFE model and the Cournot model. The simulation results show that the LSFE under (a,b)-parametrization converges to a focal equilibrium which is equal to

113

Cournot equilibrium under the capacity and transmission constraints in all cases. The simulation results have also shown that the system may not have a pure equilibrium due to the transmission congestion. But an equilibrium continuum may exist when the generation capacity of generators becomes tighter in LSFE under *a*-, *b*- and $(a \propto b)$ -parametrization. The solutions of market power analysis are different and depending on the parameters selected in different types of LSFE. It is also demonstrated that when these constraints are hit, the market price will increase and the abuse of market power can become more serious. But tighter constraints lead the solution of LSFE under *a*-, *b*- and $(a \propto b)$ -parametrization close to the solution of Cournot model.

Secondly, the parallel CGA has been employed to study market equilibrium in multi-period electricity market. The capacity constraints and inter-temporal constraints are considered in the market model. Simulation shows when ISO compels the market GenCo uses a supply function be constant across multiple pricing period, the market will generate low price than the Cournot price that can emerge in a market where a supply function is specified for a single pricing period. It is important to observe outcomes of market rule to set up an obligation for consistent bids or not.

Finally, a two stage game model is presented to investigate whether a rational GenCo would enter forward market or not only by economic incentive and what factors could affect this strategic behavior. A co-evolutionary approach is successfully employed to determine the market equilibrium in this two stage game model. Two numerical examples have been used to substantiate the theoretical analysis. Simulation results show that whether GenCos would enter the forward market relies heavily on what type competition in the spot market. When the GenCos use Cournot type competition or LSFE type competition with (a,b)-parametrization, they will enter the forward market and mitigate the market power. They also enter forward market when the capacity is hit. Otherwise they would not enter the forward market. The degree to which a GenCo enters to the forward market is also affected by the slope of the demand function and the slope of its marginal cost function.

6.2 Future Work

An emerging issue in market power analysis and market design has been the interaction of pollutant emission permits markets and energy markets. One generator (i.e. thermal unit) only allows emitting a fixed amount of pollutant, and this permit of generator can be traded in secondary markets. To make sure that an affected facility's emissions do not exceed the amount of its emission, the facility can reduce pollution through operational or equipment changes, or purchase permits from other companies who have excess permits. Excess permits can be sold, or banked for future use. If permits are in short supply and there is significant market concentration, it may be possible for large generators to exercise market power in both energy and permits markets. Profit-maximizing strategies might differ for such GenCos when both markets are considered. Then market simulation should also consider total regional or national emissions.

In Chapter 5, a two-stage game model is presented to investigate two-settlement market. One limitation of the model is the assumption of risk neutrality on the GenCos.

Unfortunately, introducing risk aversion will make the objective functions of GenCos non-quadratic which significantly increases the computational complexity of the model. And future works are also needed to investigate the bidding behaviors of GenCos when the risk aversion GenCos exist or the price difference between spot and forward markets (i.e. not perfect arbitrage) are considered. These works are needed to make the simulation results close to real market outcome.

APPENDIX

Appendix A: List of Abbreviation

ACE	Agent-based Computation Economics
ACO	Ant Colony Optimization
BBP	Bid-Based Pool
CCA	Cooperative Co-evolutionary Algorithm
CGA	Co-evolutionary Genetic Algorithm
CV	Conjectural Variation
CVBS	Conjecture Variable Based Bidding Strategy
DC	Direct Current
DE	Differential Evolution
DP	Dynamic Programming
DP-SC	Dynamic Programming-Sequential Combination
EA	Evolutionary Algorithm
ED	Economic Dispatch
EP	Evolutionary Programming
EPEC	Equilibrium Problem with Equilibrium Constraints
ES	Evolutionary Strategies
GA	Genetic Algorithm
GenCo	Generation Company
GTBS	Game Theoretic Bidding Strategies
HHI	Herfindahl–Hirschman Index
ISO	Independent System Operator
LCP	Linear Complementary Program
LI	Lerner Index
LSF	Linear Supply Function
LSFE	Linear Supply Function Equilibrium
MC	Marginal Cost
MPEC	Mathematical Program with Equilibrium Constraints

МО	Market Operator
NE	Nash Equilibrium
NLP	Nonlinear Complementary Program
OPF	Optimal Power Flow
PCMI	Price-Cost Margin Index
PSO	Particle Swarm Optimization
SFE	Supply Function Equilibrium
UC	Unit Commitment

Appendix B: Data for IEEE 30-bus system



Figure B.1: IEEE 30-bus test system

Generator No.	Bus No.	$a_i ((MW)^2h)$	<i>b</i> _{<i>i</i>} (\$/MWh)	C_i (\$/h)	$\underline{P_{Gi}}$ (MW)	$\overline{P_{Gi}}$ (MW)
1	1	0.0075	2.0	0	50	200
2	2	0.035	1.75	0	20	80
3	5	0.125	1.0	0	15	50
4	8	0.01668	3.25	0	10	35
5	11	0.05	3.0	0	10	30
6	13	0.05	3.0	0	12	40

	Bus No.	Resistance	Reactance	Transmission
Branch No.		<i>R</i> (p.u.)	<i>X</i> (p.u.)	capacity
1	1-2	0.0192	0.0575	130
2	1-3	0.0452	0.1852	130
3	2-4	0.0570	0.1737	65
4	3-4	0.0132	0.0379	130
5	2-5	0.0472	0.1938	130
6	2-6	0.0581	0.1763	65
7	4-6	0.0119	0.0414	90
8	5-7	0.0460	0.116	70
9	6-7	0.0267	0.082	130
10	6-8	0.0120	0.042	32
11	6-9	0.0000	0.208	65
12	6-10	0.0000	0.556	32
13	9-11	0.0000	0.208	65
14	9-10	0.0000	0.11	65
15	4-12	0.0000	0.256	65
16	12-13	0.0000	0.14	65
17	12-14	0.1231	0.2559	32
18	12-15	0.0662	0.1304	32
19	12-16	0.0945	0.1987	32
20	14-15	0.2210	0.1997	16
21	16-17	0.0824	0.1932	16
22	15-18	0.1070	0.2185	16
23	18-19	0.0639	0.1292	16
24	19-20	0.0340	0.068	32
25	10-20	0.0936	0.209	32
26	10-17	0.0324	0.0845	32
27	10-21	0.0348	0.0749	32
28	10-22	0.0727	0.1499	32
29	21-22	0.0116	0.0236	32
30	15-23	0.1000	0.202	16
31	22-24	0.1150	0.179	16
32	23-24	0.1320	0.27	16
33	24-25	0.1885	0.3292	16
34	25-26	0.2544	0.38	16
35	25-27	0.1093	0.2087	16
36	27-28	0.0000	0.396	65
37	27-29	0.2198	0.4153	16
38	27-30	0.3202	0.6027	16
39	29-30	0.2399	0.4533	16
40	8-28	0.0636	0.2	32
41	6-28	0.0169	0.0599	32

Table B.2: Branch data for IEEE 30-bus system

Bus No.	Load(MW)	Bus No.	Load(MW)
1	0	16	3.5
2	21.7	17	9.0
3	2.4	18	3.2
4	7.6	19	9.5
5	94.2	20	2.2
6	0.0	21	17.5
7	22.8	22	0.0
8	30.0	23	3.2
9	0.0	24	8.7
10	5.8	25	0.0
11	0.0	26	3.5
12	11.2	27	0.0
13	0.0	28	0.0
14	6.2	29	2.4
15	8.2	30	10.6

Table B.3: Load data for IEEE 30-bus system

REFERENCES

- [1]S. Stoft, Power System Economics: Designing Markets for Electricity. New York: Wiley-IEEE press, 2002.
- [2]S. Borenstein and J. Bushnell, "An empirical analysis of the potential for market power in California's electricity industry," *J. Ind. Econ.*, vol. 47, pp. 285-323, 1999.
- [3]S. Borenstein, "The trouble with electricity markets: understanding California's restructuring disaster," *The Journal of Economic Perspectives*, vol. 16, pp. 191-211, 2002.
- [4]A. K. David and F. S. Wen, "Market power in electricity supply," IEEE Trans. Energy Conversion, vol. 16, pp. 352-360, Dec 2001.
- [5]J. C. Dalton, "Assessing the competitiveness of restructured generation service markets," *Electricity Journal*, vol. 10, pp. 30-39, 1997.
- [6]S. Borenstein, J. Bushnell, and F. Wolak, "Diagnosing market power in California's deregulated wholesale electricity market," POWER working paper PWP-064, University of California Energy Institute, 1999.
- [7]P. Visudhiphan and M. D. Ilic, "Dependence of generation market power on demand/supply ratio: Analysis and modeling," in *Proceedings of IEEE Power Engineering Society 2000 Winter Meeting*, Singapore, 2000.
- [8]H. Singh, "Market power mitigation in electricity markets," in *IEEE Tutorial on Game Theory Applications in Electric Power Markets*, 1999, pp. 70-77.

- [9]M. Ventosa, A. Baillo, A. Ramos, and M. Rivier, "Electricity market modeling trends," *Energy Policy*, vol. 33, pp. 897-913, May 2005.
- [10]R. Rajamaran, L. Kirsch, F. Alvarado, and C. Clark, "Optimal selfcommitment under uncertain energy and reserve prices," in "The Next Generation of Electric Power Unit Commitment Models", Boston: Kluwer Academic Publishers, pp.93-116, 2001.
- [11]F. S. Wen and A. K. David, "Optimal bidding strategies and modeling of imperfect information among competitive generators," *IEEE Trans. Power Syst.*, vol. 16, pp. 15-21, Feb 2001.
- [12]R. Green, "Reshaping the CEGB : Electricity privatization in the UK," Utilities Policy vol. 1, pp. 245-254, 1991.
- [13]Z. Yu, F. T. Sparrow, T. L. Morin, and G. Nderitu, "A Stackelberg price leadership model with application to deregulated electricity markets," in *Proceeding of IEEE Power Engineering Society Winter Meeting*, 2000, pp. 1814-1819.
- [14]M. D. Latorre and S. Granville, "The Stackelberg equilibrium applied to AC power systems A non-interior point algorithm," *IEEE Trans. Power Syst.*, vol. 18, pp. 611-618, May 2003.
- [15] W. Hogan, "A market power model with strategic interaction in electricity networks," *Energy Journal*, vol. 18, pp. 107-141, 1997.
- [16] J. B. Cardell, C. C. Hitt, and W. W. Hogan, "Market power and strategic interaction in electricity networks," *Resource Energy Econ.*, vol. 19, pp. 109-137, 1997.

- [17] D. H. M. Nguyen and K. P. Wong, "Power markets analysis using genetic algorithm with population," in *Proceedings of PowerCon 2000 International Conference*, 2000, pp. 721-726.
- [18]J. Contreras, "Numerical Solutions to Nash-Cournot Equilibria in Coupled Constraint Electricity Markets," *IEEE Trans. Power Syst.*, vol. 19, pp. 195-206, 2004.
- [19]L. B. Cunningham, R. Baldick, and M. L. Baughman, "An empirical study of applied game theory: Transmission constrained Cournot behavior," *IEEE Trans. Power Syst.*, vol. 17, pp. 166-172, Feb 2002.
- [20] T. Peng and K. Tomsovic, "Congestion Influence on Bidding Strategies in an Electricity Market," *IEEE Trans. Power Syst.*, vol. 18, pp. 1054-1061, 2003.
- [21]K. H. Lee and R. Baldick, "Tuning of discretization in bimatrix game approach to power system market analysis," *IEEE Trans. Power Syst.*, vol. 18, pp. 830-836, May 2003.
- [22]K. H. Lee and R. Baldick, "Solving three-player games by the matrix approach with application to an electric power market," *IEEE Trans. Power Syst.*, vol. 18, pp. 1573-1580, Nov 2003.
- [23] J. Yao, I. Adler, and S. S. Oren, "Modeling and computing two-settlement oligopolistic equilibrium in a congested electricity network," *Operations Research*, vol. 56, pp. 34-47, 2008.
- [24]P. D. Klemperer and M. A. Meyer, "Supply function equilibria in oligopoly under uncertainty," *Econometrica*, vol. 57, pp. 1243-1277, 1989.

- [25] R. Green and D. M. Newbery, "Competition in the British electric spot market," J. Political Econ., vol. 100, pp. 929-953, 1992.
- [26] R. Green, "Increasing competition in the British electricity spot market," J. Indust. Econ., vol. 44, pp. 205-216, 1996.
- [27]R. J. Green, "The electricity contract market in England and Wales," J. Indust. Econ., vol. 47, pp. 107-124, 1999.
- [28] A. Rudkevich, M. Duckworth, and R. Rosen, "Modeling electricity pricing in a deregulated generation industry: The potential for oligopoly pricing in a Poolco," *Energy J.*, vol. 19, pp. 19-48, 1998.
- [29]A. Rudkevich, "Supply function equilibrium in power markets: learning all the way," TCA Technical paper, Tabors Caramanis and Associates, 1999.
- [30]R. Baldick and W. Hogan, "Capacity constrained supply function equilibrium models of electricity markets: Stability, non-decreasing constraints, and function space iterations," POWER paper, PWP-089, Univ. Calif. Energy Inst. 2001.
- [31]R. Baldick, R. Grant, and E. Kahn, "Linear supply function equilibrium: generalizations, application, and limitations," POWER paper, PWP-078, Univ. Calif. Energy Inst. 2000.
- [32]R. Baldick, R. Grant, and E. Kahn, "Theory and application of linear supply function equilibrium in electricity markets," *J. Regulatory Econ.*, vol. 25, pp. 143-167, 2004.

- [33]H. Chen, K. P. Wong, C. Y. Chung, and D. H. M. Nguyen, "A coevolutionary approach to analyzing supply function equilibrium model," *IEEE Trans. Power Syst.*, vol. 21, pp. 1019-1028, 2006.
- [34]R. Baldick, "Electricity market equilibrium models: The effect of parametrization," *IEEE Trans. Power Syst.*, vol. 17, pp. 1170-1176, 2002.
- [35]J. D. Weber and T. J. Overbye, "An individual welfare maximization algorithm for electricity markets," *IEEE Trans. Power Syst.*, vol. 17, pp. 590-596, 2002.
- [36] B. Hobbs, C. Metzler, and J. S. Pang, "Strategic gaming analysis for electric power systems: An MPEC approach," *IEEE Trans. Power Syst.*, vol. 15, pp. 638-645, 2000.
- [37]X. Wang, Y. Li, and S. Zhang, "Oligopolistic equilibrium analysis for electricity markets: a nonlinear complementarity approach," *IEEE Trans. Power Syst.*, vol. 19, pp. 1348-1355, 2004.
- [38]A. G. Alcalde, M. Ventosa, M. Rivier, A. Ramos, and G. Relan'o, "Fitting electricity market models: a conjectural variations approach," in *Proceeding of 14th Power Systems Computation Conf(PSCC)*, Seville, 2002.
- [39]C. J. Day, B. F. Hobbs, and J. S. Pang, "Oligopolistic competition in power networks: A conjectured supply function approach," *IEEE Trans. Power Syst.*, vol. 17, pp. 597-607, Aug 2002.
- [40] Y. Q. Song, Y. X. Ni, F. S. Wen, Z. J. Hou, and F. F. Wu, "Conjectural variation based bidding strategy in spot markets: fundamentals and comparison with classical game theoretical bidding strategies," *Electric Power Systems Research*, vol. 67, pp. 45-51, Oct 2003.

- [41]Y. Q. Song, Y. X. Ni, F. S. Wen, and F. F. Wu, "Conjectural variation based learning model of strategic bidding in spot market," *International Journal of Electrical Power & Energy Systems*, vol. 26, pp. 797-804, Dec 2004.
- [42]I. Otero-Novas, C. Meseguer, C. Batlle, and J. J. Alba, "A simulation model for a competitive generation market," *IEEE Trans. Power Syst.*, vol. 15, pp. 250-256, Feb 2000.
- [43]C. Day and D. Bunn, "Divestiture of generation assets in the electricity pool of England and Wales: a computational approach to analyzing market power.," *Journal* of Regulatory Economics, vol. 19, pp. 123-141, 2001.
- [44] J. Bower and D. Bunn, "Model-based comparisons of pool bilateral markets for electricity," *Energy Journal*, vol. 21, pp. 1-29, 2000.
- [45]D. J. Veit, A. Weidlich, J. Yao, and S. S.Oren, "Simulating the dynamics in two-settlement electricity markets via an agent-based approach," *International Journal of Management Science and Engineering Management*, vol. 1, pp. 83-97, 2006.
- [46]Z. Younes and M. Ilic, "Generation strategies for gaming transmission constraints: will the deregulated electric power market be an oligopoly," *Decision Support Systems*, vol. 24, pp. 207-222, 1999.
- [47] Y. Liu and F. F. Wu, "Impacts of network constraints on electricity market equilibrium," *IEEE Trans. Power Syst.*, vol. 22, pp. 126-135, 2007.

- [48]D. Bunn and F. Oliveira, "Agent-Based Simulation: An Application to the New Electricity Trading Arrangements of England and Wales," *IEEE Transactions on Evol. Comp.*, vol. 5, pp. 493-503, 2001.
- [49]D. W. Bunn and F. S. Oliveira, "Evaluating Individual Market Power in Electricity Markets via Agent-Based Simulation," *Annals of Operations Research*, vol. 121, pp. 57-77, 2003.
- [50]S.-k. Yu and J.-h. Yuan, "Agent-based Computational Economics: Methodology and Its Application in Electricity Market Research," in *Proceedings Of the 7th International Power Engineering Conference (IPEC)*, 2005.
- [51] A. J. Bagnall and G. D. Smith, "A multi-agent model of the UK market in electricity generation," *IEEE Trans. Evol. Comp.*, vol. 9, pp. 522-536, 2005.
- [52] J. Yao, S. S. Oren, and I. Adler, "Two-settlement electricity markets with price caps and Cournot generation firms," *European Journal of Operational Research*, vol. 181, pp. 1279-1296, Sep 2007.
- [53]S. Stoft, "Using game theory to study market power in simple networks," in *IEEE Tutorial Game Theory Applicat. Power Syst.*, 1999, pp. 33-40.
- [54]B. F. Hobbs, "Linear complementarity models of Nash-Cournot competition in bilateral and POOLCO power markets," *IEEE Trans. Power Syst.*, vol. 6, pp. 194-202, 2001.
- [55]Y. S. Son and R. Baldick, "Hybrid coevolutionary programming for Nash equilibrium search in games with local optima," *IEEE Trans. Evol. Comp.*, vol. 8, pp. 305-315, Aug 2004.

- [56] H. Chen, K. P. Wong, D. H. M. Nguyen, and C. Y. Chung, "Analyzing oligopolistic electricity market using coevolutionary computation," *IEEE Trans. Power Syst.*, vol. 21, pp. 143-152, 2006.
- [57]T. C. Price, "Using co-evolutionary programming to simulate strategic behavior in markets," J. Evol. Econ., vol. 7, pp. 219-254, 1997.
- [58]T. D. H. Cau and E. J. Anderson, "A co-evolutionary approach to modeling the behaviour of participants in competitive electricity markets," in *Proceeding of the IEEE Power Eng. Soc. Summer Meeting*, 2002, pp. 1534-1540.
- [59]L. Xu and Y.-X. Yu, "Transmission constrained linear supply function equilibrium in power markets: method and example," in *Proceeding of the International Conference on Power System Technology* Kunming, China, 2002, pp. 1349-1354.
- [60] D. E. Goldberg, Genetic Algorithms in Search, Optimization and Machine learning.Reading, Mass.: Addison-Wesley Pub. Co., 1989.
- [61]H.-G. Beyer and H.-P. Schwefel, "Evolution strategies: A comprehensive introduction," *Natural Computing*, vol. 1, pp. 3-52, 2002.
- [62] A. E. Eiben and J. E. Smith, Introduction to Evolutionary Computing. Berlin: Springer, 2003.
- [63] J. Kennedy and R. Eberhart, "Particle swarm optimization," in IEEE Int. Conf. on Neural Networks, Piscataway, NJ, 1995, pp. 1942-1948.
- [64]I. Trelea, "The particle swarm optimization algorithm: convergence analysis and parameter selection," *Information Processing Letters*, vol. 85, pp. 317-325, 2003.

- [65]R. P. Wiegand, "An analysis of cooperative coevolutionary algorithms," Ph.D. thesis, Fairfax: George Mason Univ, 2003.
- [66]C. Rosin and R. Belew, "New methods for competitive coevolution," *Evolutionary Computation*, vol. 5, pp. 1-29, 1996.
- [67]S. Luke, C. Hohn, J. Farris, G. Jackson, and J. Hendler, "Co-evolving soccer softbot team coordination with genetic programming," RoboCup-97: Robot Soccer World Cup I (Lecture Notes in AI No. 1395), Berlin 1998.
- [68]M. Potter and K. D. Jong, "A cooperative coevolutionary approach to function optimization," in *Proceedings of the Third Conference on Parallel Problem Solving from Nature* Jerusalem, Israel, 1994.
- [69] R. P. Wiegand, "Applying diffusion to a cooperative coevolutionary model," in Proceedings of the Fifth International Conference on Parallel Problem Solving from Nature (PPSN V),, 1998, pp. 560-569.
- [70] M. Potter, "The Design and Analysis of a Computational Model of Cooperative CoEvolution," Ph.D. thesis, Fairfax, Virginia.: George Mason University, 1997.
- [71]H. Y. Chen and X. F. Wang, "Cooperative Coevolutionary Algorithm for unit commitment," *IEEE Trans. Power Syst.*, vol. 17, pp. 128-133, Feb 2002.
- [72]C. H. Liang, C. Y. Chung, K. P. Wong, and X. Z. Duan, "Parallel optimal reactive power flow based on cooperative co-evolutionary differential evolution and power system decomposition," *IEEE Trans. Power Syst.*, vol. 22, pp. 249-257, Feb 2007.
- [73]M. Sefrioui and J. Periaux, "Nash genetic algorithms: examples and applications," in *Proceedings of the 2000 Congress on Evolutionary Computation CEC*, 2000, pp. 509-516.
- [74]R. P. Wiegand, W. Liles, and K. D. Jong, "An empirical analysis of collaboration methods in cooperative coevolutionary algorithms," in *Proceedings of the Genetic* and Evolutionary Computation Conference, Morgan Kaufmann, 2001.
- [75]R. P. Wiegand, W. C. Liles, and K. A. De Jong, "Analyzing cooperative coevolution with evolutionary game theory," in *Proceedings of the 2002 Congressn* on Evolutionary Computation, 2002, pp. 1600-1605.
- [76]L. Panait and S. Luke, "Time-dependent collaboration schemes for cooperative coevolutionary algorithms," in *Proceedings of the AAAI Fall Symposium on Coevolutionary and Coadaptive Systems*, 2005.
- [77]E. Popovici and K. D. Jong, "A dynamical systems analysis of collaboration methods in cooperative co-evolution," in *Proceedings of the AAAI Fall Symposium on Coevolutionary and Coadaptive Systems*, 2005.
- [78]E. Popovici and K. D. Jong, "Understanding cooperative coevolutionary dynamics via simple fitness landscapes," in *Proceedings of the Genetic and Evolutionary Computation Conference*, New York, 2005.
- [79]E. Popovici and K. A. De Jong, "The dynamics of the best individuals in co-evolution," *Natural Computing*, vol. 5, pp. 229-255, 2006.

- [80]T. Jansen and R. P. Wiegand, "Sequential versus parallel cooperative coevolutionary (1+1) EAs," in *Proceeding of the 2003 Congress on Evolutionary Computation*, 2003, pp. 30-37.
- [81]E. Popovici and K. A. De Jong, "Sequential versus Parallel Cooperative Coevolutionary for Optimization," in *Proceeding of IEEE Congress on Evolutionary Computation*, Vancouver, BC, Canada, 2006, pp. 1610-1617.
- [82] M. A. Potter and K. A. De Jong, "Cooperative coevolution: an architecture for evolving coadapted subcomponents," *Evol. Comput.*, vol. 8, pp. 1-29, 2000.
- [83]T. Jansen and R. P. Wiegand, "The cooperative coevolutionary (1+1) EA," *Evolutionary Computation*, vol. 12, pp. 405-434, 2004.
- [84] D. Fudenberg and J. Tirole, *Game Theory*. Cambridge, Mass.: MIT Press, 1991.
- [85]G. Ellison, "Learning, local interaction and coordination," *Econometrica*, vol. 61, pp. 1047-1071, 1993.
- [86] M. Kandori, G. Mailath, and R. Rob, "Learning, mutation and long-run equilibrium in games," *Econometrica*, vol. 61, pp. 1105-1125, 1993.
- [87]X. Wang, "Equilibrium analysis of electricity markets using nonlinear complementarity methods," Ph.D. dissertation, Shanghai, China: Shanghai Univ., 2006.
- [88]T. Riechmann, "Genetic algorithm learning and evolutionary games," Econ. Dynamics Contr., vol. 25, pp. 1019-1037, 2001.

- [89]F. Gao, G. Gutierrez-Alcaraz, and G. B. Sheble, "Comparison of Artificial Life Techniques for Market Simulation," in *Proceedings of the 39th Hawaii International Conference on System Sciences*, 2006.
- [90] D. H. M. Nguyen and K. P. Wong, "Natural dynamic equilibrium and multiple equilibria of competitive power markets," *IEE Proc. -Gener. Transm. Distrib.*, vol. 149, pp. 133-138, 2002.
- [91]R. Baldick and W. Hogan, "Stability of supply function equilibria: implications for daily vs. hourly bids in a poolco market," *Journal of Regulatory Economics*, vol. 30, pp. 119-139, 2006.
- [92]C. A. Berry and B. F. Hobbs, "Analyzing Strategic Bidding Behavior in Transmission Networks," *Utilities Policy*, vol. 8, pp. 139-158, 1999.
- [93] R. Ivanic, P. Preckel, and Z. Yu, "Market power and welfare effects in DC power flow electricity models with thermal losses," *Decision Support System* vol. 40, 2005.
- [94] J. D. Weber and T. J. Overbye, "A two-level optimization problem for analysis of market bidding strategies," in *Proceeding of the IEEE Power Eng. Soc. Summer Meeting*, Edmonton, AB, Canada, 1999, pp. 682-687.
- [95]J. Tirole, *The Theory of Industrial Organization*. Cambridge, Mass.: MIT Press, 1988.
- [96]O. Alsac and B. Stott, "Optimal load flow with steady-state security," *IEEE Trans. Power Appar. Syst.*, vol. 93, pp. 745-751, 1973.

- [97]B. F. Hobbs, M. H. Rothkopf, R. P. O'Neill, and H.-p. Chao, *The Next Generation of Electric Power Unit Commitment Models* Boston: Kluwer Academic Publishers, 2001.
- [98]S. d. laTorre, A. J. Conejo, and J. Contreras, "Simulating oligopolistic pool-based electricity markets: a multiperiod approach," *IEEE Trans. Power Syst.*, vol. 18, pp. 1547-1555, 2003.
- [99]R. Garca-Bertrand, A. J. Conejo, and S. Gabriel, "Electricity market near-equilibrium under locational marginal pricing and minimum profit conditions," *European Journal of Operational Research*, pp. 457-479, 2005
- [100] C. K. Pang and H. C. Chen, "Optimal short term thermal unit commitment," *IEEE Trans. on PAS*, vol. 95, pp. 1336-1346, 1976.
- [101] L. Tesfatsion, "Agent-based computational economics growing economies from the bottom up," <u>http://www.econ.iastate.edu/tesfatsi/ace.htm</u>, Economics Working Paper No. 1, Iowa State University, 2002.
- [102] F. Herrera, M. Lozano, and J. L. Verdegay, "Tackling Real-Coded genetic algorithms: operators and tools for behavioural analysis," *Artificial Intelligence Review*, vol. 12, pp. 265-319, 1998.
- [103] I. J. Raglend and N. P. Padhy, "Solutions to practical unit commitment problems with operational, power flow and environmental constraints," in *Proceeding of the Power Engineering Society General Meeting*, 2006

- [104] M. Liu and F. F. Wu, "Risk management in a competitive electricity market," *International Journal of Electrical Power & Energy Systems*, vol. 29, pp. 690-697, Nov 2007.
- [105] B. Allaz, "Oligopoly, uncertainty and strategic forward transactions," *Inter. J. Indu. Organ.*, vol. 10, pp. 297-308, 1992.
- [106] B. Allaz and J. Vila, "Cournot competition, forward markets and efficiency," J. *Econ. Theory*, vol. 59, pp. 1-16, 1993.
- [107] D. M. Newbery, "Competition, contracts and entry in the electricity spot market," *Rand J. Econ.*, vol. 29, pp. 726-749, 1998.
- [108] H. Niu, R. Baldick, and G. D. Zhu, "Supply function equilibrium bidding strategies with fixed forward contracts," *IEEE Trans. Power Syst.*, vol. 20, pp. 1859-1867, Nov 2005.
- [109] T. S. Chung, S. H. Zhang, K. P. Wong, C. W. Yu, and C. Y. Chung, "Strategic forward contracting in electricity markets: modelling and analysis by equilibrium method," *IEE Proc. -Gener. Transm. Distrib.*, vol. 151, pp. 141-149, 2004.
- [110] H. Moore, *MATLAB for Engineers*. Upper Saddle River, NJ: Pearson/Prentice Hall, 2007.